Good for the Environment, Good for Business: Foreign Acquisitions and Energy Intensity

Arlan Brucal^{*} Beata Javorcik[†] Inessa Love[‡]

This version: June 2019

Abstract

The link between foreign ownership and environmental performance remains a controversial issue. This paper contributes to our understanding of this subject by analyzing the impact of foreign acquisitions on plant-level energy intensity. The analysis applies a difference-indifferences approach combined with propensity score matching to the data from the Indonesian Manufacturing Census for the period 1983-2001 (or 1983-2008 in robustness checks). It covers 210 acquisition cases where an acquired plant is observed two years before and at least three years after an ownership change and for which a carefully selected control plant exists. The results suggest that while foreign ownership increases the overall energy usage due to expansion of output, it decreases the plant's energy intensity. Specifically, acquired plants reduce energy intensity by about 30% two years after acquisition, relative to the control plants. In contrast, foreign divestments tend to increase energy intensity. At the aggregate level, entry of foreign-owned plants is associated with industry-wide reduction in energy intensity.

^{*}Grantham Research Institute on Climate Change and the Environment, London School of Economics and Political Science, Houghton Street, London WC2A 2AE, U.K. E-mail: a.z.brucal@lse.ac.uk. Brucal acknowledges support from the Grantham Foundation and the Economic and Social Research Council (ESRC) through the Centre for Climate Change Economics and Policy. This paper has also received funding from the European Union's Horizon 2020 research and innovation programme under grant agreement No 770680.

[†]Department of Economics, University of Oxford and CEPR, Manor Road Building, Manor Road, Oxford OX1 3UQ, UK. E-mail: beata.javorcik@economics.ox.ac.uk.

[‡]Department of Economics, University of Hawaii at Manoa, 2424 Maile Way, Saunders Hall 542, Honolulu, HI 96822, USA. E-mail: ilove@hawaii.edu.

1 Introduction

Foreign direct investment (FDI) has been a powerful force of convergence across countries. In addition to bringing capital and creating jobs, FDI stimulates economic growth by enhancing firm-level efficiency. It does so directly by transferring cutting-edge technologies and management practices to its affiliates (Arnold and Javorcik, 2009; Chen, 2011; Javorcik and Poelhekke, 2017) and encouraging product and process innovation (Guadalupe et al., 2012) as well as indirectly through knowledge spillovers (Javorcik, 2004; Havranek and Irsova, 2011).

The spectacular growth in FDI flows, along with the increasing importance of developing economies as host countries, has raised concerns about the potential effect of FDI on the natural environment (Zarsky, 1999). On the one hand, environmentalists argue that highly polluting multinationals relocate to countries with weaker environmental standards in order to circumvent costly regulations in their home country (Hanna, 2010; Millimet and Roy, 2015; Cai et al., 2016). In this way, they increase pollution levels not only in host countries but also globally.¹ On the other hand, supporters of globalization point to studies that fail to find evidence of multinationals in polluting industries being attracted to locations with weak regulations (Dean et al., 2009; Javorcik and Wei, 2004) and point out that FDI may have a positive effect on natural environment because multinationals tend to use more advanced technologies and production methods than their domestic counterparts.² Since the existing literature (reviewed below) has produced mixed results, the issue remains controversial (Kellenberg, 2009; Cole et al., 2017).

This study contributes to our understanding of the link between FDI and environmental protection by taking a novel approach. Rather than examining whether FDI flows are influenced by environmental standards in the destination countries or whether polluting industries are more likely to engage in FDI, we examine the impact of foreign acquisitions on energy consumption and carbon dioxide (CO₂) emissions of acquired plants.³ We use plant-level panel data from the Indonesian Manufacturing Census covering the period 1983-2001 (or 1983-2008 in a robustness check). To investigate the impact of foreign acquisitions on plant performance, we combine a difference-indifferences approach with propensity score matching, where matching is done within industry-year cells. This allows us to account for selection on observables and unobservable time-invariant plant heterogeneity and for confounding factors that affect both domestic and foreign-owned establish-

¹Anecdotal evidence abound. For instance, in 2013, the smoke haze from Indonesia's palm oil production, which was dominated by foreign investors, elevated to dangerous levels and caused significant health hazards not only in Indonesia, but also in Malaysia and Singapore (Chachavalpongpun, 2013).

²Rondinelli and Berry (2000) list a number of examples showcasing how multinationals and their affiliates help improve the environmental condition of their host country. Blackman and Wu (1999) use survey results to demonstrate that FDI plants outperform their Chinese counterparts in terms of energy efficiency, largely due the use of advanced efficiency-enhancing generation technologies.

 $^{^{3}}$ A foreign acquisition takes place when a firm headquartered abroad buys a significant stake (of at least 10%) in a domestic firm in order to assume partial or full control over it. This case is distinct from 'greenfield investment'– a foreign direct investment where a parent company establishes an entirely new facility.

ments within the same industry in the same year. To the best of our knowledge, this study is the first to employ such an approach in determining the effect of foreign ownership on plant-level environmental performance.

The ideal measure of a plant's environmental performance is the total amount of pollution the plant emits at a particular time period, principally because it accounts for (potentially) different pollution abatement technologies applied by different firms. However, getting data on plant-level emissions across periods remains extremely difficult, particularly in developing countries. This is the reason why a number of studies resort to approximations of environmental performance using plant-level expenditures on pollution-emitting inputs such as energy (see, for example, Eskeland and Harrison, 2003; Cole et al., 2008a; Barrows and Ollivier, 2014, 2018). Building upon this literature, we approximate plant-level environmental performance by considering energy usage in physical units and converting it into CO_2 emissions using standard conversion factors specific to each type of energy input.

We can observe fuel switching because the dataset includes plant-level expenditures (in Rps) and physical usage (e.g., in metric tons or liters) of each energy input. The energy inputs consist of fuels and lubricants and electricity. Fuels and lubricants are divided into more detailed inputs, which include gasoline, diesel, diesel oil, kerosene, lubricant, bunker oil, coal, coke, public gas, liquefied petroleum gas (LPG), firewood, and charcoal. We have information on the amount of fuels and lubricants that are used for electricity generation, as some of the plants produce electricity for their own consumption and for sale to other end users. With this information, any reallocation to lower-carbon inputs is captured in our emission measure.⁴

Our analysis is based on 210 foreign acquisition cases where an acquired plant is observed two years before and at least three years after an ownership change and for which a carefully selected control plant exists. The results suggest that, while foreign ownership increases total energy use and carbon dioxide (CO_2) emissions in acquired plants due to expansion of the production scale, it lowers the energy and emission intensity of output. The reduction in energy use relative to output is nontrivial, ranging from 26% in the acquisition year to 30% two years later. These results are robust to different matching and estimation procedures, a longer time horizon, accounting for the potential effect of markups, and taking into account competitive pressures from foreign affiliates within the same local market. They are also robust to extending the sample time period to 1983-

⁴A global pollutant, such as CO_2 , may be of lesser interest compared to local pollutants (e.g., particulate matter and sulfur dioxide). Nonetheless, CO_2 is extremely difficult to abate. This was particularly true in Indonesia during our sample period, which would make our energy and emissions measures a close approximation of total CO_2 emissions. To date, there are three kinds of fossil-fuel-based carbon abatement technologies: (1) higher efficiency conversion processes; (2) fuel switching to lower carbon alternatives; and (3) carbon capture and storage (CCS). Differences in combustion efficiency are already captured by differences in plant-level total energy usage, on which we have information. It is also unlikely that the CCS technology was available in Indonesia during our sample period, as the first wave of feasibility studies for CCS in the country were conducted in early 2003 to 2005 (Best et al., 2011).

2008. While this extension allows us to consider more acquisition cases, it comes at the price of less disaggregated data on fuel usage, which is why we focus on the shorter time period in the main analysis.

We also find that the reduction in fuel intensity takes place immediately after the ownership change, while the reduction in electricity intensity happens more slowly and is somewhat less pronounced. Our results also indicate that plants with different initial energy intensity benefit from acquisition differently. In particular, plants with higher energy intensity (possibly smaller and less efficient plants) tend to reduce their energy and emission intensities more than those that are already less energy intensive. This finding might explain why previous literature on the relationship between foreign ownership and plant-level energy intensity produced mixed results (see, for example, Eskeland and Harrison, 2003; Cole et al., 2008a).

To shed light on the channels through which foreign ownership leads to improvements in energy efficiency we bring in an additional dataset and demonstrate that our conclusions hold even if we restrict attention to plants with next to no changes in the output mix and when we control for contemporaneous changes in output. In other words, we eliminate the possibility that foreign ownership works solely through changes to the production structure or the scale channel. Our findings are thus very suggestive of foreign acquisitions being associated with improvements to the production process taking place through introduction of better technologies and better management.

In an additional exercise, which is quite novel relative to the existing literature, we consider foreign divestments, i.e., sales of foreign affiliates to domestic owners. We find that such divestments are accompanied by an increase in energy and emission intensities as well as a decline in output. The increase in energy use relative to output is quite substantial, reaching 29% two years after the ownership change. This finding is consistent with the conclusions of Javorcik and Poelhekke (2017) who show deterioration in performance after foreign divestments and conclude that the productivity advantage associated with foreign ownership result from continuous injections of knowledge and management practices from the parent company.

In a motivating exercise conducted at the aggregate level, we find that energy and emission intensities in Indonesian manufacturing as a whole improved by 31% from 1983 to 2001. We show that at the industry level the decline in the aggregate weighted energy intensity is positively associated with the increased presence of foreign affiliates. The improvement seems to be driven by both within-plant reduction in energy intensity as well as reallocation of market shares towards more energy-efficient producers.⁵

⁵As we employ data on FDI inflows to Indonesia, our results capture the impact of FDI flows from (mostly) developed countries to a developing economy. While there is no reason to believe that FDI from developing regions to developed regions would reduce energy intensity, it would be interesting to know if energy intensity would also decline when FDI come from developing countries with about the same level of environmental stringency. Our dataset, unfortunately, does not provide information on source countries to shed light on this issue.

Our paper contributes to the literature examining how foreign ownership influences plant-level environmental performance. Within this broad literature, there are very few papers that study firm's actual energy use or pollution emissions, and the evidence is still mixed. Pargal and Wheeler (1996) use information on plant-level emissions of water pollution, measured in terms of biological oxygen demand (i.e., kilograms of oxygen needed over 5 days to completely oxidize the organic pollutants emitted), which was collected by the Indonesian Environment Ministry's PROKASIH (Clean Rivers) program for 1989-90. After controlling for plant scale, age and efficiency, they find that foreign ownership does not have a significant effect on water pollution emissions. Their results rely on cross-sectional data and thus capture correlations rather than a causal effect. Eskeland and Harrison (2003) examine plant-level data from Cote d'Ivoire, Mexico and Venezuela and find that the energy share, i.e., the cost of energy use divided by the total value of the plant's output, is negatively related to foreign ownership. Although the authors control for plant demand for other inputs and some plant characteristics, data limitations prevent them from controlling for plant fixed effects. Their results should be interpreted as correlations. Cole et al. (2008a) use plant-level data from Ghana and find no strong evidence of foreign ownership influencing total energy use. Instead, and perhaps correlated with foreign ownership, plants with foreign-trained managers are found to have lower energy intensity. Again, the study is unable to take into account unobservable plant heterogeneity and captures correlations. Albornoz et al. (2009) employ a crosssection of approximately 1,200 firms in Argentina and find a positive correlation between foreign ownership and implementation of environmental management systems.⁶

Our paper extends this literature in several ways. First, we take into account selection into foreign ownership, i.e., the possibility that foreign investors choose to acquire local plants with better environmental performance, and thus our results come much closer to capturing a causal effect of foreign ownership. Second, we work with panel data and hence we are able to take into account unobservable plant heterogeneity and examine the stability of the estimated effects over time. Third, we are also able to measure energy use in physical units and provide a more detailed analysis on the types of fuel used.

Our work also makes a contribution to the relatively new literature examining the effects of foreign acquisitions on the acquired plants. This literature relies on propensity score matching combined with a difference-in-differences approach (Arnold and Javorcik, 2009; Chen, 2011; Wang and Wang, 2015; Bircan, 2019) or uses inverse probability of treatment weighting (Guadalupe et al., 2012) to address selection into foreign ownership. We use the former methodology, which is well established in this literature, but focus on a completely different outcome. While the existing studies focus

⁶In a related study, Cole et al. (2011) consider per-capita emissions of pollutants in 112 major Chinese cities during the 2001-4 period. They find that the share of output of foreign-owned firms increases emissions while output of firms from Hong Kong, Macao, and Taiwan either reduces pollution or has no effect. The aggregated nature of the data (city level) makes it difficult to draw conclusions about the effect of foreign ownership on firm-level environmental performance.

on plant-level total factor productivity, export and import intensity, profitability and innovation, we aim to capture plant-level environmental performance by examining plant-level energy intensity and CO_2 emission intensity. These outcomes have not been considered by the existing studies.

The rest of the paper is organized as follows. Section 2 provides background information on Indonesia and on why multinational owners are more likely to invest in energy efficiency. It also describes how we measure energy usage and emissions. Section 3 presents motivation for our analysis: an industry-level exercise demonstrating correlations between energy efficiency and the presence of foreign owned plants. Section 4 discusses our empirical approach to understanding the link between FDI and plant-level energy intensity. In section 5, we present the main results from our analysis. Section 6 deepens the analysis by looking at changes in use of individual energy inputs, channels through which foreign ownership may affect energy efficiency and nonlinear effect arising from different pre-acquisition energy intensity. Section 7 examines the effects of foreign divestments. Section 8 concludes by considering potential policy implications of our findings.

2 Background Information

2.1 Overview of Indonesia

Indonesia is a suitable setting for studying the effects of foreign acquisitions on plant-level energy efficiency. This is for two reasons. First, the country received significant inflows of FDI, ranking as the 5^{th} largest recipient of FDI among developing countries in the mid 1990s (see Arnold and Javorcik, 2009). The influx of FDI was in part driven by significant reductions in trade barriers and industrial deregulation in the early 1980s and 1990-1996. The period of high capital inflows also coincided with economic recessions in the United States, Japan and some European countries. During this period, competing FDI destinations, such as Thailand, experienced a rise in labor cost and had limited infrastructure, which may have prompted investors to reallocate their portfolios to other emerging economies, including Indonesia. Second, as pointed out by Garcia et al. (2007), environmental protection in Indonesia was generally weak and ineffective during the country's impressive industrial growth in the 1980s and 1990s. Thus, the results of our analysis will not be influenced by pollution-related policies in the host country and will give us a cleaner picture of the impact of FDI on plant performance in terms of energy efficiency.

Figure 1 illustrates Indonesia's net FDI inflows, Gross Domestic Product (GDP), CO_2 emissions and the number of foreign-owned plants from 1983 to 2001. The vertical bars represent the net FDI inflows from balance-of-payment statistics measured in million US\$, the solid line indicates the country's GDP in 100 million US\$, the dashed line is the number of foreign-owned plants (calculated based on the Census of Manufacturing) and the dotted line is the CO_2 emissions in 100,000 metric tons (based on Oak Ridge National Laboratory data). The graph shows an upward trend in all the variables, with the number of foreign plants picking up in 1984, followed by the GDP and net FDI flows increasing around 1987. Interestingly, the GDP increased at a much faster pace than did emissions, suggesting that emissions per dollar unit of output may have declined during the period. Both the GDP and FDI inflows experienced a significant decline following the 1997-1998 Asian financial crisis.





Sources of Data: The World Bank, Oak Ridge National Laboratory, Indonesian Census of Manufacturing

2.2 Why would we expect foreign acquisitions to improve energy efficiency?

Improvements in energy use amongst plants can be achieved through different methods, ranging from those that call for little or no investment to obtain immediate paybacks to those that would require a sizable commitment of capital funds. The low- or no-cost opportunities include installing energy-saving lighting options, influencing employee behavior by keeping employees informed about the plants' energy-efficiency goals and progress and recognizing them for their role in supporting these initiatives, and partnering with utility providers to identify appropriate procurement or demand-side management plans. Other channels that might entail some capital costs include introducing new technology (e.g., reducing process-heating costs by introducing waste-heat recovery technology) and improving mechanical performance by altering operation procedures of machinery or replacing old machinery with more energy efficient one. Plants may also implement environmental/energy management standards (e.g., ISO 14001). Other channels, which may require significant capital costs, include improving facility design, continuous research and development or hiring energy managers.

There are several reasons why foreign acquisitions may improve energy efficiency. First, foreign

acquisitions tend to increase production volume by boosting productivity and facilitating access to foreign markets through the distribution network of the foreign parent (Arnold and Javorcik (2009)). This makes investments in improving energy efficiency more worthwhile as the sales base becomes large enough to cover the fixed cost of investment.⁷

Second, investments in energy efficiency amongst multinationals may be driven by factors inherent to companies involved in a global supply chain. For example, multinationals based in OECD countries may employ more energy-efficient and cleaner technologies in compliance with more stringent regulations or standards implemented in the region, compared with other companies in developing countries (Cole et al., 2008b). The use of these energy-efficient technologies and management practices may be passed on to their affiliates in developing countries to maintain their production standards and meet the requirements of their environmentally conscious export markets. These technologies and management systems can be also indirectly passed on to the affiliates' suppliers to maintain their global standards and reputation.

Third, investment in energy efficiency fundamentally involves decisions on higher initial capital costs and uncertain lower future energy costs at present values (Gillingham et al., 2009). Consequently, firm-level characteristics can be crucial in determining a firm's propensity to invest in improving energy efficiency (DeCanio and Watkins, 1998). Some domestic firms may under-invest in energy efficient technologies due to capital constraints or financial issues (Anderson and Newell, 2004). Information problems, such as shortsightedness and bounded rationality of management, may also force locally owned firms to resort to sub-optimal alternatives (DeCanio, 1993, 1998). In contrast, these issues are less serious for foreign affiliates as they generally dominate locally-owned firms in terms of investment (Arnold and Javorcik, 2009) and in international training and experience of decision makers within the firm (Cole et al., 2008a).

Fourth, superior management practices are widely believed to be a characteristic of multinational companies. And better management practices make it easier to introduce low-cost efficiency improvements described at the beginning of this subsection.

Anecdotal evidence supports the view that foreign affiliates tend to be more energy efficient than local plants. For instance, Byrne et al. (2014) report that multinational cement companies in Sub-Saharan Africa are more energy efficient than locally owned companies producing mainly for their local markets. The authors also find that locally owned firms have poorer access to knowledge on low-carbon technologies and have weaker incentives to innovate.

⁷This type of argument has been made by (Guadalupe et al., 2012) in the context of foreign ownership encouraging product and process innovation, and by (Barrows and Ollivier, 2014; Forslid et al., 2018) in the context of exporting encouraging better environmental performance.

2.3 Measuring energy usage and emissions

We use data from *Survei Manufaktur*, the Indonesian Census of Manufacturing conducted by the National Statistical Office (BPS). The data encompass all manufacturing plants with 20 or more employees on an annual basis since 1975. The census has detailed information on fuel and electricity use, both in terms of values and physical quantities.⁸ The sample available to us spans the period from 1983 to 2001 covering about 40,000 plants with 300,400 plant-year observations.

The main advantage of our data is the availability of detailed information on plant-level expenditures and physical usage (e.g., in metric tons, kWh or liters) of each energy input. The energy inputs consist of fuels and lubricants and electricity. Fuels and lubricants are divided into more detailed inputs, which include: gasoline, diesel, diesel oil, kerosene, lubricant, bunker oil, coal, coke, public gas, liquefied petroleum gas (LPG), firewood, and charcoal. We also have information on the amount of fuel and lubricants used for electricity generation, as some of the plants produce electricity for their own consumption and for sale to other end users. The detailed information on the kind of energy inputs used will allow us to capture any reallocation to lower-carbon inputs that may be associated with ownership changes.

Our data set also includes information on the amount of electricity sold and the amount bought from the state-owned power company *Perusahaan Listrik Negara* (PLN) and from other independent power producers (non-PLN). When calculating the total energy usage, we use total electricity purchased less electricity sold. Recognizing that some of the plants were generating their own electricity, we also account for the total amount of fuel used to generate their own electricity.

The original data set includes 300,400 plant-year observations. 61,561 of these contain positive energy expenditure on a particular energy input but do not include the information on energy use in physical units. We impute the plant-level physical energy consumption using the following equation:

$$lny_{it} = \beta lnCost_{it} + ISIC_k + PROV_l + YEAR_m + u_{it}$$
⁽¹⁾

where y_{it} is the physical measure of energy inputs (e.g., gasoline in liters) and $Cost_{it}$ is total cost of energy inputs (e.g., gasoline in '000 Rps). *ISIC*, *PROV* and *YEAR* are industry (ISIC 4-digit level), province and year fixed effects, respectively. We estimate this specification separately for each energy input.⁹ In this way, we impute the usage of each energy input in physical units. Except for kerosene, the model explains more than 90% of the variation of energy use in physical units.¹⁰

The energy and emission content of each energy input (in British Thermal Units or BTUs) is

⁸The survey questionnaires and other relevant information about the dataset can be accessed online at http: //www.rand.org/labor/bps/statistik_industri.html.

 $^{{}^{9}}$ Table B.1 reports the list of energy inputs as well as the goodness-of-fit from estimating equation 1.

¹⁰There are also observations where there are physical units but total cost is missing. We regard these observations as missing in the data.

calculated using conversion factors from reliable US agencies and institutions found in Table B.2.¹¹¹² When calculating CO_2 emissions, we take into account the mix of source fuels used to generate electricity in Indonesia. Namely, we used the conversion factors for electricity generated from bituminous and sub-bituminous coal – the primary source of electricity in the country.

The key variables of interest are energy intensity (defined in two ways) and emission intensity. Following Eskeland and Harrison (2003), we first define energy intensity as total expenditure on energy per 1000 Rps of real output. It is plausible that acquired firms, possibly due to greater reliance on international markets, use technologies that utilize more expensive but cleaner energy inputs. Thus, energy expenditure per unit of output may remain constant (or even increase) but actual energy use (in BTUs) per unit of output may decline. In order to address this issue, we also express energy intensity as the total physical energy use (in MBTUs) per currency unit of output. We repeat the same process to calculate a plant's emission intensity, which is defined as total carbon dioxide emissions (in kg CO_2) per currency unit of output.

3 Motivating Exercise at the Industry Level

To motivate our plant-level analysis we first describe the developments in the aggregate energy intensity for the entire Indonesian manufacturing sector during our sample period. We decompose these developments into changes in the unweighted average energy intensity and reallocation effects. Then we examine how aggregate energy and the two components are affected by the number of foreign affiliates participating in the local industry.

Following Olley and Pakes (1996) and Pavcnik (2002), we compile the aggregate energy intensity measure W_t , which is the average of the plants' individual energy intensities (i.e., energy expenditure/output) weighted by the plant's share in total manufacturing output s_{it} . We calculate W_t for the entire Indonesian manufacturing sector for each year t. Then we decompose the aggregate energy intensity into the unweighted aggregate energy intensity (i.e., the average energy intensity taken over all plants) and the covariance between plant's share of the entire sector's output and its

100 barrels diesel x
$$\frac{5.825 \text{ million BTUS (MBTUS)}}{1 \text{ barrel}} = 582.50 \text{ MBTUS}$$

We follow the same procedure for calculating CO_2 emissions (in kg CO_2). Using the same example above, we calculate the CO_2 emissions as below:

582.50 MBTUs x
$$\frac{71.80 \text{ kg CO}_2}{1 \text{ MBTU}} = 41,845.04 \text{ kg CO}_2$$

Note that the conversion factors are time-invariant, which implicitly assumes that abatement cost is constant over time. Foreign firms may have resorted to more efficient conversion processes over time, which is not observed in the data.

¹¹British Thermal Unit (BTU) is a traditional unit of energy. The US Energy Information Administration interprets BTU as the amount of energy needed to heat one pound of water from 39 to 40 degrees Fahrenheit (EIA, 2011).

 $^{^{12}}$ A sample calculation of the energy usage of a plant using 100 barrels of diesel fuel at a certain time period is illustrated below:

energy intensity:

$$\underbrace{W_t = \sum_{i} s_{it} \ln EIP_{it}}_{\text{Aggregate weighted}} = \underbrace{\overline{lnEIP}_t}_{\text{Unweighted average}} + \underbrace{\sum_{i} (s_{it} - \overline{s}_t)(lnEIP_{it} - \overline{lnEIP}_t)}_{Covariance} \tag{2}$$

where s_{it} is the share of plant *i*'s output in the total manufacturing output at time t, \overline{s}_t is the average output share, $lnEIP_{it}$ is firm *i*'s log(energy expenditure/output), \overline{lnEIP}_t is the average log(energy expenditure/output) over all plants in the sector.

A change in the first term (unweighted average energy intensity) captures within-plant improvements in energy intensity. The second term (covariance), if positive, indicates that more output is produced by more energy intensive producers. A change in the covariance captures the effects of reallocation of market shares and resources across firms with different energy intensity levels.

Our calculations show that aggregate energy intensity for the entire Indonesian manufacturing sector has declined by 31% in 2001 relative to the 1983 levels. We also find that the aggregate energy intensity is negatively associated with increases in foreign affiliates, particularly before the 1997 financial crisis, and net FDI inflows (see Figure 2).

Given the strong downward trend of aggregate energy intensity as more foreign affiliates enter the market, it is natural to ask how changes in the number of participating foreign affiliates are associated with industry-wide aggregate energy intensity. To answer this question, we regress aggregate energy intensity and each of its components on the number of foreign affiliates in a particular industry. More specifically, we calculate the aggregate weighted energy intensity, unweighted average energy intensity and covariance at the 4-digit ISIC level. Then following Harrison et al. (2012) and Javorcik and Li (2013), we estimate the following equation:

$$Y_{jst} = \beta ForeignAffiliates_{jt} + \gamma_j + \lambda_{st} + \varepsilon_{jst}$$

$$\tag{3}$$

where Y_{jst} is the aggregate energy intensity and its components relevant to 4-digit ISIC industry j operating in 2-digit ISIC sector s in year t and ForeignAffiliates is the log-transformed number of foreign affiliates in the industry.¹³ γ_j and λ_{st} are industry and sector-year fixed effects, respectively. We weight all observations using the maximum number of plants observed in each industry during the entire sample period to ensure that industries with large plant populations receive higher weight, which makes our result representative of the national level. We cluster standard errors at

 $^{^{13}}$ A foreign affiliate is defined as a plant with at least 20% of foreign equity. As some industry-year cells have no foreign affiliates, we add 1 to the number of foreign affiliates before transforming it into logarithm.





Note: Aggregate energy intensity figures are relative to 1983 levels. Source: No. of Foreign Affiliates (Indonesian Census of Manufacturing); Aggregate energy intensity measures (Authors' calculation); Net FDI Inflows (The World Bank).

the industry level.

To test the robustness of our result, we repeat the above regression using the share of foreign affiliates in the industry output as our indicator of foreign affiliates' participation in the market. We also look at the different aggregate measures of environmental performance, including energy and CO_2 emission intensities normalized either by output or by expenditure on materials (the latter to check for potential markup effects). Table 1 presents our results.

Our estimation results imply that participation of foreign affiliates is negatively associated with industry-level aggregate energy and emission intensities. This suggests that foreign affiliates' participation may be facilitating improvements in aggregate measures of environmental performance. We find that both within-plant improvement and reallocation towards bigger and less energy-intensive plants drive the improvement in aggregate energy intensities, though the impact on reallocation is not robust to measuring foreign affiliate presence in terms of output shares.

Table 1: Regression results:	Decomposition	of weighted	aggregate	energy
	$\mathbf{intensity}$			

	Measure h	based on nun	nber of FAs	Measure	based on outp	out share of FAs
	W_t	\overline{lnEIP}	Covariance	$ W_t$	\overline{lnEIP}	Covariance
Log (Energy Expenditure/Output)						
Foreign Affiliates	-0.226***	-0.086**	-0.140***	-0.772*	-0.552**	-0.219
-	(0.041)	(0.034)	(0.044)	(0.410)	(0.276)	(0.318)
Adj. R-sq.	0.853	0.829	0.774	0.842	0.827	0.764
Observations	1408	1408	1408	1408	1408	1408
Log (Energy Use/Output)						
Foreign Affiliates	-0.215***	-0.070**	-0.146***	-0.740*	-0.490*	-0.250
	(0.039)	(0.034)	(0.040)	(0.401)	(0.271)	(0.336)
Adi. R-sa.	0.859	0.852	0.784	0.850	0.851	0.775
Observations	1408	1408	1408	1408	1408	1408
Log (CO2 Emissions/Output)						
Foreign Affiliates	-0.217***	-0.077**	-0.140***	-0.761*	-0.521*	-0.239
	(0.039)	(0.035)	(0.040)	(0.405)	(0.277)	(0.328)
Adi, R-sa.	0.853	0.834	0.783	0.844	0.833	0.775
Observations	1408	1408	1408	1408	1408	1408
Log (Energy Expenditure/Materials)						
Foreign Affiliates	-0.254^{***}	-0.093**	-0.161**	-0.822*	-0.698**	-0.125
	(0.061)	(0.043)	(0.065)	(0.454)	(0.307)	(0.381)
Adj. R-sq.	0.873	0.874	0.789	0.863	0.874	0.779
Observations	1407	1407	1407	1407	1407	1407
Log (Energy Use/Materials)						
Foreign Affiliates	-0.243***	-0.076*	-0.167^{***}	-0.788*	-0.628**	-0.160
0	(0.058)	(0.041)	(0.060)	(0.444)	(0.299)	(0.392)
Adj. R-sq.	0.881	0.882	0.804	0.872	0.883	0.794
Observations	1407	1407	1407	1407	1407	1407
Log (CO2 Emissions/Materials)						
Foreign Affiliates	-0.244***	-0.082**	-0.162^{***}	-0.804*	-0.657**	-0.147
0	(0.057)	(0.041)	(0.060)	(0.450)	(0.301)	(0.386)
Adj. R-sq.	0.877	0.872	0.805	0.868	0.873	0.796
Observations	1407	1407	1407	1407	1407	1407
No. of industries (4-digit ISIC)	79	79	79	79	79	79
No. of sectors (2-digit ISIC)	9	9	9	9	9	9
No. of years	19	19	19	19	19	19

Note: Period coverage is 1983-2001. Each regression includes 4-digit ISIC industry and 2-digit ISIC-year fixed effects. FA stands for foreign affiliates. Robust standard errors clustered at the 4-digit ISIC industry level are in parentheses. *, **, *** indicate statistical significance at 0.10, 0.05, and 0.01 level, respectively.

4 Main Analysis: Empirical Strategy

4.1 Identifying foreign acquisitions

Following Arnold and Javorcik (2009), we define a foreign acquisition as a change in the foreign ownership share from less than 20% to at least 20%. The exact value of the threshold does not affect our results because more than 99% of future acquisition targets have a foreign capital share equal to zero in the pre-acquisition period. And in 95% of the cases, the post-acquisition foreign ownership shares is at least 25%. In more than 75% of cases, it is at least 50% (Figure 3).

Acquired plants are distributed across a wide range of industries, but mostly concentrated in energy-intensive sectors. The main one is manufacturing of fabricated metal products, machinery and equipment which comprises 25.6% of the acquired plants in the sample data. 21.8% are involved in textile and wearing apparel industries, and 19.1% are in manufacture of chemicals and chemical, petroleum, coal, rubber and plastic products.¹⁴

The first three columns of Table 2 provides summary statistics for plants that have gone through a foreign acquisition and for plants that have always remained domestic. Acquired plants outperform domestic plants in terms of almost all the economic variables. For example, they are on average much bigger, employ a higher share of skilled labor, rely more on international markets and invest more in machinery. In terms of environmental attributes, they spend more on energy in absolute terms and emit more CO_2 . Their production process is, however, less energy intensive and produces lower CO_2 emissions per unit of output.



Figure 3: Distribution of foreign ownership in acquired firms, 1983-2001.

Source: Indonesian Census of Manufacturing

 $^{^{14}}$ For more description on the distribution of acquired plants by industry and acquisition cases by year, see Figures A.1 and A.2 in the Appendix, respectively.

The empirical strategy used to identify the effect of foreign ownership on plant-level energy efficiency is anchored on three foundations. First, the study focuses on the changes from domestic to foreign ownership taking place within the same plant. In particular, we consider plants that are observed for at least five consecutive years and which have initially less than 20% foreign equity and at least 20% of equity belonging to foreign owners thereafter.¹⁵ By focusing on ownership change we are able to take into account the selection bias that would plague a comparison of domestic plants to all foreign plants, due to the possibility that foreign affiliates may choose the most energy-efficient domestic plants. This approach, however, dramatically reduces the number of observations that can be considered. Fortunately, thanks to large FDI inflows into Indonesia during the sample period, we are able to observe enough acquisition cases to allow us to generalize the results with confidence.

Second, we use a difference-in-differences approach to compare the performance of foreign-acquired plants with the performance of plants remaining in domestic hands. This approach eliminates the influence of all unobservable elements of the acquisition decision that are constant or strongly persistent over time.

The main challenge is how to develop a reasonable estimate of the counterfactual, that is, the change in the variables of interest that would have been observed had the acquisition not occurred. It is well recognized that the estimates obtained by comparing a treatment group with the remainder of the population could be biased if the two groups have significant differences in pre-treatment observable characteristics (Dehejia and Wahba, 2002). As shown in the left panel of Table 2, the domestic plants are different from acquired plants even before they were acquired in almost all the aspects, including their output and energy consumption, suggesting that running a simple difference-in-differences may not yield an unbiased estimate of the impact of foreign acquisition on plant-level environmental performance.

Thus, third, in order to develop a reasonable counterfactual, we employ a one-to-one propensity score matching (PSM) (Rosenbaum and Rubin, 1983, 1985). That is, for each treated plant that will be acquired by foreign investors next period we identify a control plant with similar characteristics using the procedure developed by Leuven and Sianesi (2012). We ensure that each acquired plant is paired with a domestic plant that is operating in the same sector and year and has very similar output and energy intensity in each of the two years preceding the ownership change.¹⁶

¹⁵We focus only on the first ownership change within plant that meets these criteria. We are unable to observe whether the same foreign firm acquired multiple domestic plants.

¹⁶See Table 2 for a list of variables used in matching. Austin (2011) argues that there is a lack of consensus in the applied literature as to which variables should be included in the propensity score. There is merit in including only those that affect the outcome, and there is merit in including those that affect both the outcome and the treatment (Austin et al., 2007). However, a larger number of included variables, while deemed safe, tends to result in fewer matched pairs. This will be particularly true in our case, as there are many missing observations in certain baseline covariates. Therefore, we chose the variables that are strongly correlated with the outcome variables and which could bring the highest number of matched pairs while ensuring balance.

Matching within the industry-year cell ensures that we control for sector- and time-specific confounding factors that affect both domestic and acquired firms. We use a matching procedure with replacement. In robustness checks, we will also employ other methods, such as, coarsened exact matching (Iacus et al., 2011) and inverse propensity to treatment weighting (Hirano and Imbens, 2001).

The underlying assumption for the validity of the procedure is that conditional on the observable characteristics that are relevant for the acquisition decision, potential outcomes for the treated and control plants are orthogonal to treatment status. Thus, we need to argue that both the treated and the control plants most likely faced the same business and regulatory environments, energy prices, sector-specific and macroeconomic shocks, and trends before the acquisition taking place. This is partly established by matching within the same industry-year cell. When we compare the sample means of the variables used in matching procedure between the treatment group (210 acquired plants) and the control group (210 domestic plants), we find that there is no statistically significant difference in the pre-acquisition period (see columns labeled "Matched sample" in the top panel of Table 2). It is even more comforting that, except for the share of imported materials, there is no significant difference between the means of variables not used in matching procedure has performed quite well in obtaining control units that are comparable to acquired plants on nearly all observed covariates.

We also assess whether our matched acquired plants are representative of all foreign acquisitions found in the data that fulfill the minimum criteria for data completeness. First, 210 matched plants comprise about 40% of 555 acquired plants in the sample. Second, the distribution of the 210 acquired plants across industries is roughly the same as that of the all acquired plants, except for the textile industry (see Figure A.3). This is because the textile industry is dominated by foreign-owned plants which makes it difficult to find a suitable domestic control plant for all acquisition cases within the same industry-year cell and within our specified caliper. Third, we find no statistically significant difference between the matched acquired plants and all acquisition cases in terms of the key variables used in matching, such as, energy expenditure/output at t - 1 and t - 2 and output at t-2 (see the right hand side panel in Table 2). There is also no statistically significant difference in terms of pre-acquisition employment, exporter status, share of imported inputs, CO_2 emission intensity as well as all the pre-acquisition trends considered. However, the matched acquisitions seem to operate at a somewhat larger scale in terms of output and hence total energy use and emissions in the pre-acquisition period.

)				•	4		
	D	matched d	ata	Ma	tched sam	ple	Matched v	s unmatched	Acquisitions
Variables	(555 acqui	ired vs 39,653	2 domestic)	(210 trea	ted vs 210	controls)	(210 match	ed vs 555 all	acquisitions)
	Acquired	Domestic	p-value	Treated	Controls	p-value	Matched	All	p-value
Trood in motohing							·	Acquisitions	
Used in muching Log (Output)t-1	9.55	7.74	0.000	9.89	9.88	0.951	9.89	9.55	0.024
Log (Energy expenditure/output)t-1	-3.98	-3.82	0.005	-3.87	-3.83	0.752	-3.87	-3.98	0.302
Log (Output)t-2	9.82	7.79	0.000	9.74	9.74	0.997	9.74	9.82	0.602
Log (Energy expenditure/output)t-2	-3.93	-3.81	0.132	-3.93	-3.86	0.574	-3.93	-3.93	0.973
Unused in matching									
Log (Energy expenditure)t-1	5.59	4.00	0.000	6.02	6.04	0.868	6.02	5.59	0.005
Log(Energy use)t-1	8.51	6.96	0.000	8.95	9.00	0.779	8.95	8.51	0.005
Log (CO2 emissions)t-1	12.90	11.35	0.000	13.33	13.38	0.760	13.33	12.90	0.007
Log (Employment)t-1	5.04	4.12	0.000	5.18	5.29	0.338	5.18	5.04	0.143
Exporter dummy t-1	0.21	0.08	0.000	0.19	0.18	0.706	0.19	0.21	0.646
Share of imported materials t-1	0.27	0.08	0.000	0.26	0.19	0.050	0.26	0.27	0.641
Share of skilled workers t-1	0.21	0.14	0.000	0.24	0.22	0.291	0.24	0.21	0.034
Log(Investment in machinery)t-1	7.65	5.46	0.000	8.19	7.80	0.105	8.19	7.65	0.015
Log(Energy use/output)t-1	-1.06	-0.86	0.001	-0.94	-0.87	0.645	-0.94	-1.06	0.252
Log(CO2 emissions/output)t-1	3.33	3.53	0.001	3.44	3.50	0.612	3.44	3.33	0.332
Log(Energy exp./materials exp.)t-1	-3.08	-2.98	0.188	-2.87	-3.07	0.201	-2.87	-3.08	0.111
Δ Log (Output)t-1	0.19	0.05	0.000	0.15	0.14	0.893	0.15	0.19	0.578
Δ Log (Energy expenditure)t-1	0.17	0.06	0.032	0.21	0.17	0.644	0.21	0.17	0.619
Δ Log (Energy use)t-1	0.17	0.07	0.080	0.22	0.20	0.887	0.22	0.17	0.587
Δ Log (CO2 emissions)t-1	0.17	0.08	0.102	0.21	0.20	0.857	0.21	0.17	0.591
Δ Log (Energy expenditure/output)t-1	-0.02	0.01	0.593	0.06	0.03	0.684	0.06	-0.02	0.307
Δ Log(Energy use/output)t-1	-0.02	0.03	0.465	0.06	0.06	0.975	0.06	-0.02	0.296
$\Delta \operatorname{Log}(\operatorname{CO2} \text{emissions/output})t-1$	-0.02	0.03	0.408	0.06	0.05	0.938	0.06	-0.02	0.299
Δ Log(Energy exp./materials exp.)t-1	-0.03	0.01	0.550	0.02	0.04	0.842	0.02	-0.03	0.549
Source: Indonesian Census of Manu	ufacturing								

Table 2: Balancing test for matched domestic and acquired plants.

After obtaining the matched pairs, we examine the effect of foreign acquisitions on outcomes of interest using a difference-in-differences approach. More specifically, we estimate the following equation on the matched sample which is observed in the pre-acquisition period and in one of the post-acquisition periods:

$$y_{it} = \alpha_i + \gamma Post_t + \beta (Post_t * Acquired_i) + \varepsilon_{it}$$

$$\tag{4}$$

where *i* denotes plant and *t* is the year. We compare two periods, i.e., t = T - 1, T + s where *T* is the acquisition year and s = 0,1,2. A separate model is estimated for each *s*.

5 Results

This section presents the main results of the paper. We start by analyzing the influence of foreign acquisition on selected outcome variables in a simple OLS framework, ignoring the potential selection bias. Next, we present the results from estimating equation 4 using our matched sample. We then address a number of potential other concerns and perform a series of robustness checks to test the stability of our estimates.

5.1 OLS results

Before we go into the matching results, we perform a difference-in-differences estimation on the unmatched sample ignoring the selection bias and controlling only for 4-digit ISIC-industry-year fixed effects. The results, presented in Panel A of Table 3, show that acquired plants experience a large, persistent and statistically significant increase in output, accompanied by an increase in energy use (both in terms of expenditure and physical units) and CO_2 emissions. Meanwhile, there is also a substantial, persistent and statistically significant decline in energy cost of output and in CO_2 emissions per unit of output.¹⁷

In Panel B, we additionally control for unobserved time-invariant plant-level heterogeneity. Doing so effectively controls for pre-acquisition characteristics of acquired affiliates. The effect on the variables of interests remains the same, except that the magnitude of effect becomes smaller. This pattern suggests that there may be some unobserved non-random plant-level characteristics associated with acquisition decisions that are influencing the outcome variables. The results indicate that it is important to address selection bias in the analysis.

¹⁷The energy and emission content of each energy input (in British Thermal Units or BTUs) is calculated using conversion factors from reliable US agencies and institutions found in Table B.2. A sample calculation is found in footnote 12.

		Panel (A)			Panel (B)			Panel (C)	
	Always doi	mestic + acquir	ed plants	Always dc	mestic + acqui	red plants	Always domest	iic + matched a	cquired plants
	t	t+1	t+2	t	t+1	t+2	t	t+1	t+2
Log(Output) 2.	156^{***}	2.339^{***}	2.423^{***}	1.034^{***}	1.197^{***}	1.245^{***}	0.839^{***}	0.974^{***}	0.949^{***}
Log(Energy Expenditure) 1.	***022	1.938^{***}	1.985^{***}	0.766^{***}	0.906^{***}	0.916^{***}	0.647^{***}	0.763^{***}	0.703^{***}
Log(Energy Use, in MBTUs) 1.	724***	1.913^{***}	1.967^{***}	0.712^{***}	0.868^{***}	0.891^{***}	0.584^{***}	0.721^{***}	0.703^{***}
Log (CO ₂ Emissions) 1.	743***	1.927^{***}	1.971^{***}	0.727^{***}	0.878^{***}	0.890^{***}	0.621^{***}	0.746^{***}	0.721^{***}
Log(Energy Expenditure/Output) -0.	.347***	-0.373***	-0.414^{***}	-0.249***	-0.269^{***}	-0.323***	-0.194^{**}	-0.214^{**}	-0.266^{***}
Log(Energy Use/Output) -0.	.393***	-0.398***	-0.431^{***}	-0.302***	-0.307^{***}	-0.348***	-0.257^{***}	-0.256^{***}	-0.267^{**}
Log (CO ₂ Emissions/Output) -0.	$.374^{***}$	-0.384***	-0.428^{***}	-0.287***	-0.297***	-0.349^{***}	-0.220**	-0.230***	-0.248^{**}
Log(Energy Expenditure/Materials) -0.	.331***	-0.333***	-0.412^{***}	-0.295***	-0.288***	-0.395^{***}	-0.252^{**}	-0.217^{**}	-0.384^{***}
Log(Energy Use/Materials) -0.	$.369^{***}$	-0.357***	-0.433^{***}	-0.342***	-0.327^{***}	-0.418^{***}	-0.308***	-0.264^{***}	-0.378***
Log (CO ₂ Emissions/Materials) -0.	.350***	-0.342***	-0.427***	-0.330***	-0.321***	-0.423^{***}	-0.280***	-0.245^{**}	-0.370***
Industry-year fixed effects	Yes	Yes	γ_{es}	Yes	Yes	Yes	$\mathbf{Y}_{\mathbf{es}}$	\mathbf{Yes}	Yes
Firm fixed effects	N_{O}	No	No	Yes	Yes	Yes	\mathbf{Yes}	Yes	Yes
Sample Size	16	39,141 - 180,901		1	65,905 - 177,840		1	65,307 - 177,15	10

Table 3: Difference-in-differences on the unmatched sample.

each column) and a year before foreign-owned plants were acquired. All panels include plants that have always been domestic. Panels A and B also include all domestic plants that have undergone foreign acquisitions. In Panel C only those acquired plants that are used in matching regressions are included. Heteroskedasticity-robust standard errors clustered at the plant level are in parentheses. *, **, *** indicate statistical significance at 0.10, 0.05, and 0.01 Each entry corresponds to a separate regression for a particular outcome variable covering two time periods: the year relative to acquisition (as listed in ١Ž

level, respectively.

In Panel (C), we repeat the exercise from Panel (B) dropping the acquisition cases that are not included in our matched sample as described in Section 4. Doing so makes the estimated effects slightly smaller and somewhat less significant. This improves our confidence in the estimates because it indicates that there is little evidence of sample selection when it comes to which acquisition cases are included in the final matching exercise. If anything, focusing on the smaller sample will lead us to underestimate the effects of foreign acquisition on the variables of interest.

5.2 Results from the Difference-in-Differences on the Matched Sample

Perhaps the most transparent and intuitive way of presenting the impact of acquisition on plantlevel environmental performance is through a graph where the outcomes for the matched domestic and acquired plants are placed side-by-side before and after the acquisition (see Figure 4). There is a number of features of the graph that are worth discussing. First, it appears that the matching procedure created a set of domestic and acquired plants that are very comparable to each other prior to the time of acquisition. Both groups display very similar paths two years prior to the ownership change. This comparability holds true across all variables of interest. Second, the paths start to diverge already in the acquisition year and the gap between the groups increases over the two subsequent years. Third, the figure suggests an increase in the use of energy, both in terms of value and physical units, and in CO_2 emission levels of the acquired plants. The increase in energy use and emissions is intuitive because these variables are positively associated with the expansion of output. However, acquired plants experience a substantial decrease in their energy consumption and emissions per unit of output relative to the plants that remain in domestic hands. The difference between energy and emission intensities of acquired plants and those of the control group gets larger over time.

The effect of foreign acquisitions is formally tested by estimating equation 4, and the results are presented in Table 4. The results suggest that the acquired plants experience a statistically significant increase in output relative to the control group already during the acquisition period.¹⁸ The effect is sizeable as the acquired plants outperform the controls by 83.8 log points or about 131%.¹⁹ The difference between the two groups increases to 101.3 log points or about 175% in the subsequent two years. This pattern is consistent with the findings of Arnold and Javorcik (2009) who looked at the effect on foreign acquisitions on the scale of production.

The expansion in output coincides with significant increases in energy consumption, both in terms of expenditure and physical units (measured in monetary terms or in MBTUs, respectively), although

¹⁸One of the reasons for this immediate effect is that foreign firms transplant their management practices to host countries (Bloom et al., 2012), and that just within months improvements in management practices translate into better performance (Bloom et al., 2013).

¹⁹The change in the variable of interest associated with the foreign acquisition dummy is calculated as $e^{(\beta)} - 1$. In this case, the change is $e^{(0.838)} - 1 = 1.311$.



Figure 4: Trajectories of output, energy expenditure and energy intensities: Matched acquired vs. domestic plants

The figure illustrates the average value of each variable of interest in a given time period for the treated (acquired) and control (domestic) group. The horizontal axes indicate the year relative to the period t where treated plants are acquired.

	Acquisition Year	1 Year Later	2 Years Later
		Log(Output)	
Post*Acquired	0.838^{***}	1.047^{***}	1.013^{***}
	(0.113)	(0.117)	(0.122)
R-sq. (within)	0.203	0.240	0.229
No. of Obs.	840	840	840
	Log (Ener	rgy Expenditure	e in Rps)
Post*Acquired	0.567***	0.773***	0.705***
	(0.118)	(0.126)	(0.132)
R-sq. (within)	0.145	0.178	0.163
No. of Obs.	838	838	835
	Log(Er	nergy Use in ME	BTUs)
Post*Acquired	0.539***	0.770***	0.664***
-	(0.118)	(0.130)	(0.136)
R-sq. (within)	0.138	0.178	0.168
No. of Obs.	838	838	835
	Log	$(CO_2 Emission)$	ns)
Post*Acquired	0.562***	0.792***	0.673***
-	(0.120)	(0.130)	(0.137)
R-sq. (within)	0.150	0.188	0.176
No. of Obs.	838	838	835
	Log (Ener	gy Expenditure,	/Output)
Post*Acquired	-0.276**	-0.282**	-0.326**
	(0.119)	(0.118)	(0.127)
R-sq. (within)	0.013	0.014	0.016
No. of Obs.	838	838	835
	Log (1	Energy Use/Out	(put)
Post*Acquired	-0.304**	-0.285**	-0.367***
	(0.120)	(0.125)	(0.137)
R-sq. (within)	0.015	0.014	0.019
No. of Obs.	838	838	835
	Log (CO	D ₂ Emissions/O	utput)
Post*Acquired	-0.282**	-0.262**	-0.357***
1	(0.119)	(0.124)	(0.136)
R-sq. (within)	0.014	0.015	0.021
No. of Obs.	838	838	835

Table 4: Difference-in-differences analysis on the matched sample.

Note: The table shows the result of estimating equation 4 on the matched sample described in Section 4. The dependent variables are as listed in each panel. Each column in each panel is a separate regression for a particular outcome variable covering two time periods: the year relative to acquisition (as listed in each column) and a year before foreign affiliates were acquired. Heteroskedasticity-robust standard errors clustered at the plant level are in parentheses. *Post* dummy and plant fixed effects are included in all specifications. *, **, *** indicate statistical significance at 0.10, 0.05, and 0.01 level, respectively. the latter effect is of a lesser magnitude. For example, energy expenditure increases by about 76% during the year of acquisition while the physical energy use goes up by 71%. CO_2 emissions follow the same trend although the estimated magnitude are smaller. It should also be noted that the increases in energy and emission levels are substantially smaller when compared to increases in output.

Next we focus on energy and emission intensities, which are obtained by dividing energy consumption (measured in monetary terms and in MTBUs) and CO_2 emissions by the total value of output. The estimated magnitudes are economically meaningful. They suggest that the share of energy cost in the total value of output falls in the acquired plants (relative to the control group) by 24% in the year after the acquisition with a further decline to 28% after two years. In terms of physical energy use, the acquired plants reduce their energy intensity by 26% during the acquisition year and by about 30% two years after. We find a similar pattern for CO_2 emission intensity. Overall, the results suggest that acquired plants tend to use less energy intensive and perhaps cleaner production techniques. The finding that CO_2 emission intensity and energy intensity decline at about the same rate suggests that the observed reductions in CO_2 emission intensity are merely a by-product of the acquired firms' effort to reduce overall energy use for privately appropriable reasons.

Below we subject our findings to a series of robustness checks.

5.3 Are our findings a result of increased local competition from foreign affiliates?

One may be concerned that increased competition resulting from entry of foreign affiliates may be influencing our results. For instance, acquired plants may increase competitive pressures on the control group by seizing some of their market share. This can lead to less energy efficient production processes in the control plants due to a smaller scale. Failure to account for this effect might lead us to overestimate the effects of foreign acquisitions on energy and CO_2 intensity.

To address this concern, we follow Javorcik and Poelhekke (2017) and adjust the matching procedure so that the matched acquired and domestic plants are located in different counties ("*kabu-paten*"). In this way, we avoid the potential effects of competition in the local market which may confound our results. The estimates are presented in Table 5. They are very similar to our baseline findings. The reduction in energy and emission intensities remain negative and statistically significant.

	Acquisition Year	1 Year Later	2 Years Later
Post*Acquired R-sq. (within) No. of Obs.	0.829*** (0.114) 0.199 836	Log(Output) 1.037*** (0.116) 0.238 836	1.008*** (0.123) 0.225 836
	Log (Ene	rgy Expenditure	in Rps)
Post*Acquired	0.573*** (0.118)	0.758^{***} (0.126)	0.701^{***} (0.134)
R-sq. (within) No. of Obs.	0.145 834	0.173 834	0.161 831
	Log (E	nergy Use in MI	BTUs)
Post*Acquired	0.546^{***}	0.769^{***}	0.651^{***}
	(0.119)	(0.131)	(0.138)
R-sq. (within)	0.137	0.176	0.165
No. of Obs.	834	834	831
	Log (CO)	2 Emissions in k	$(g CO_2)$
Post*Acquired	0.560^{***}	0.785^{***}	0.653^{***}
	(0.120)	(0.132)	(0.140)
R-sq. (within)	0.150	0.186	0.174
No. of Obs.	834	834	831
	Log (Ener	gy Expenditure	/Output)
Post*Acquired	-0.262**	-0.286**	-0.324**
	(0.119)	(0.119)	(0.128)
R-sq. (within)	0.012	0.015	0.016
No. of Obs.	834	834	831
	Log (Energy Use/Out	(put)
Post*Acquired	-0.288**	-0.276**	-0.374***
-	(0.120)	(0.126)	(0.139)
R-sq. (within)	0.014	0.014	0.021
No. of Obs.	834	834	831
	Log (C	$O_2 \text{ Emissions}/O$	utput)
Post*Acquired	-0.275**	-0.259**	-0.372***
	(0.119)	(0.124)	(0.138)
R-sq. (within)	0.013	0.016	0.024
No. of Obs.	834	834	831

Table 5: Difference-in-differences analysis on the matched sample: Accounting for potential spillovers of foreign affiliates to local competitors.

Note: The table shows the result of estimating equation 4 on the matched sample described in Section 4. Matched plants do not come from the same county or *kabupaten*. The dependent variables are as listed in each panel. Each column in each panel is a separate regression for a particular outcome variable covering two time periods: the year relative to acquisition (as listed in each column) and a year before foreign affiliates were acquired. Heteroskedasticityrobust standard errors clustered at the plant level are in parentheses. *Post* dummy and plant fixed effects are included in all specifications. *, **, *** indicate statistical significance at 0.10, 0.05, and 0.01 level, respectively.

5.4 Are our findings driven by increases in markups and pre-existing global value chain relationship?

It is possible that Indonesian producers acquired by foreign investors increase their markups and that higher markups are responsible for our finding of a lower energy intensity.²⁰ Although it would not invalidate our results, as after all energy usage per unit value of output is a meaningful outcome, it is still interesting to shed some light on this issue. Therefore, we normalize energy expenditure, physical energy use and emission level by the value of expenditures on material inputs. If our result is an artifact of changes in markups, then we would not expect to see foreign acquisitions affecting these redefined energy-intensity measures.

The results, presented in the top three panels of Table 6, show that our initial findings are robust to the alternative definition of energy and emission intensities. They show that acquired plants experience a reduction in energy costs per unit cost of materials relative to domestic plants. We also see similar declines in physical energy use and CO_2 emissions per unit cost of materials. The estimated decline is actually slightly larger than when output normalization was used.

Next we check whether pre-acquisition participation in global value chains influences the extent to which foreign acquisitions lead to increased energy efficiency. To do so, we interact *Post* * *Acquired* with the pre-acquisition export share. Doing so has no impact on the acquisition effect, and the additional term is not statistically significant. Similarly, we interact *Post* * *Acquired* with pre-acquisition share of imported inputs. The additional interaction terms is not statistically significant, while the overall effect of acquisitions increases in magnitude.

5.5 Robustness Checks

In this section, we subject our results to other sensitivity and robustness checks. In particular, we test whether our results hold if we exclude the period of the Asian financial crisis, include a longer time horizon, or employ a different matching procedures.

5.5.1 Excluding the Asian financial crisis

Another possible concern is that the effect of the 1997-1998 Asian financial crisis may be influencing our results. Some studies find that, thanks to access to financing from parent companies, foreign affiliates performed significantly better than domestic firms during the crisis (see, among others, Blalock et al., 2008; Alfaro and Chen, 2012). If that's the case, then the smaller scale of production in domestic establishments resulting in lower energy efficiency could be driving our findings. Failure to account for this effect might lead to biased estimates of the effect of foreign acquisition on output and consequently on energy consumption and CO_2 emissions.

²⁰Javorcik and Poelhekke (2017) show that changes in ownership are associated with changes in markups.

	Acquisition Year	1 Year Later	2 Years Later
	Log(Energy Expe	nditure/Materia	ls Expenditure)
Post*Acquired	-0.310**	-0.266**	-0.382**
	(0.123)	(0.128)	(0.147)
R-sq. (within)	0.021	0.011	0.018
No. of Obs.	808	810	807
	Log(Energy U	Jse/Materials Ex	xpenditure)
Post*Acquired	-0.339***	-0.279**	-0.426***
	(0.124)	(0.134)	(0.153)
R-sq. (within)	0.024	0.011	0.019
No. of Obs.	808	810	807
	$Log(CO_2 \text{ Emiss})$	sions/Materials	Expenditure)
Post*Acquired	-0.328***	-0.266**	-0.428***
	(0.124)	(0.134)	(0.153)
R-sq. (within)	0.020	0.010	0.020
No. of Obs.	808	810	807
	Log(Energ	y Expenditure/	Output)
Post*Acquired	-0.275**	-0.250*	-0.358**
	(0.130)	(0.128)	(0.140)
Post*Acquired*Pre-acquisition export share	-0.000	-0.003	0.002
	(0.004)	(0.004)	(0.004)
R-sq. (within)	0.014	0.016	0.018
No. of Obs.	838	838	835
	Log(Energ	y Expenditure/	Output)
Post*Acquired	-0.331**	-0.309**	-0.433***
	(0.152)	(0.152)	(0.154)
$\operatorname{Post}^{*}\operatorname{Acquired}^{*}\operatorname{Pre-acquisition}$ share of Imported materials	0.154	0.053	0.363
	(0.330)	(0.316)	(0.366)
R-sq. (within)	0.019	0.018	0.024
No. of Obs.	838	838	835

Table 6: Difference-in-differences analysis on the matched sample: Accounting for potential markup and GVC effects.

Note: The table reflects the result of estimating equation 4 on the matched sample described in Section 4. The dependent variables are as listed in each panel. Each column in each panel is a separate regression for a particular outcome variable covering two time periods: the year relative to acquisition (as listed in each column) and a year before foreign-owned plants were acquired. Heteroskedasticity-robust standard errors are in parentheses. *Post* dummy and plant fixed effects are included in all specifications. Panels 4 and 5 also include *Post*Pre-acquisition* export share and *Post*Pre-acquisition share of imported materials*, respectively. *, **, *** indicate statistical significance at 0.10, 0.05, and 0.01 level, respectively.

To address this concern, we dropped years beyond 1997 to avoid the effects of the Asian financial crisis which may confound our results. A graph showing the unconditional mean of the variables of interest after dropping the Asian crisis period and beyond for the matched treated and control plants are presented in Figure C.1. Pre-acquisition parallel trends in all outcome variables are still observed, as well as the divergence in the acquisition year and two subsequent years. Meanwhile, the estimates are presented in Table C.1. We still observe significant reduction in energy intensity during the acquisition year and two years after. The effects for the year following the acquisition, though still negative, are not statistically significant.

5.5.2 Longer time horizon

Our analysis so far has focused on a relatively short period and, due to missing values, the number of observations fluctuated between time periods. To further check the robustness of our findings, we extend the time horizon to six years under foreign ownership. This exacerbates the problem of missing values to a large extent because we now have fewer post-treatment observations for acquisitions taking place towards the end of the sample period. To maintain comparability across periods, we keep the sample constant; that is, we focus only on plants that have non-missing observations in all periods considered for all the outcome variables.

Table C.2 present the summary of the results. While the number of observations inevitably decreases, our result remain robust and consistent with our baseline estimation. Even if we control for the potential effect of increased local competition from foreign affiliates; that is, we restrict the control group to those plants that are situated in different countries or *kabupatens*, our results remain consistent with our baseline findings (see Table C.3).

5.5.3 Are our results dependent on a particular matching procedure?

We also test the robustness of our results to different matching procedures. In order to show this, we employ a one-to-one coarsened exact matching (CEM) procedure following Iacus et al. (2009) and Iacus et al. (2011).²¹ A detailed procedure is presented in Appendix Appendix D and the results summarized in Table D.2. The results are qualitatively the same, with output, energy use and emissions increasing up to two years after the acquisition, while energy and emission intensities being significantly reduced. The treatment effects are notably larger in magnitude and more precisely estimated.

Another potential concern is that a one-to-one matching procedure requires us to drop many ob-

²¹The application of CEM is not without precedence. For example, Wang et al. (2010) use CEM to estimate the magnitude of spillovers generated by academic "superstars" to their collaborators' publication rates. Singh and Agrawal (2011) use the same matching technique to determine how firms are making use of their recruits' prior stock of ideas. Beatty and Tuttle (2015) employ CEM in studying the effect of the unprecedented increase in Supplemental Nutrition Assistance Program on participants' expenditure on food.

servations. To address this concern, we use the propensity score from the previous matching procedures to reweight observations in our difference-in-differences estimation (as, for instance, done by Guadalupe et al. (2012)). A detailed procedure is presented in Appendix E and the results summarized in Table E.1. Our results remain consistent with baseline estimates, suggesting that are findings are robust to changing the empirical approach.

5.5.4 Do our results hold in more recent data?

Another potential concern is the absence of more recent plant-level information, which may cast doubt on the applicability of our findings to a more recent time period. To address this concern, we extended our dataset up to 2008. Although this results in a larger number of acquisition cases considered, it comes at the price of much less disaggregated information on energy inputs. This is due to the fact that the post-2001 surveys aggregate some important energy inputs (such as, natural gas) into the "other" category. This makes it impossible to estimate the total energy use in physical units and to convert it into CO_2 emissions. Although one could potentially ignore the natural gas usage, doing so would bias the estimates because the average share of natural gas in total energy use had been increasing since 1983. Moreover, other factors, such as, the 2008 global financial crisis, might confound our estimates for the more recent years.

Nevertheless, with these limitations in mind, we conduct several robustness checks using the extended dataset. The results confirm that foreign acquisition increase output much more than they increase energy expenditure, resulting in a decline in energy intensity. This effect is already visible in the acquisition year and remains statistically significant in the following five years (see Table E.2). Normalizing energy expenditure by cost of materials, thus controlling for potential changes in markups, does not affect these conclusions.²²

In sum, we conclude that our findings can be generalized to a more recent time period.

6 Structural Change and Heterogeneous Effects

In this section, we deepen our analysis by looking at whether acquired plants undergo structural changes in their production processes, considering channels through which foreign ownership may affect energy efficiency and testing whether there exist heterogeneous effects among target plants with different pre-acquisition energy intensities.

 $^{^{22}}$ The balancing test for the extended data can be found in Table F.1. We fail to reject the null hypothesis that the mean difference between the domestic and acquired plants is zero for the variables we used in matching as well for the variables related to energy intensity (i.e., lagged energy intensity levels and trends).

6.1 Do acquired plants reallocate across energy inputs?

Do acquired plants change their mix of energy inputs? We investigate this question in Table F.3 which considers as outcome variables the plants' expenditures on major energy sources: fuel and lubricants, electricity (net of own generation), and their respective intensities. The left panel corresponds to results using expenditure-based measures, while in the right panel we consider energy consumption measured in physical units.

The results indicate that foreign ownership affects the plant's mix of energy sources. During the acquisition year, expenditures on fuel and lubricants in acquired plants increase by 39% relative to the control group with the gap increasing to almost 72% two years after. Meanwhile, consumption of electricity in acquired plants increases at a much higher rate, beginning at 114% during the acquisition period, then increases to 117% a year later before declining at 100% in the subsequent year. All of these changes are statistically significant. The results using energy use in physical units are almost identical and consistently significant.

As for energy intensity, we observe a significant reduction in the cost of fuel and lubricants per unit of output during the acquisition period. The acquired plants' fuel intensity declines by 34% during the acquisition period and by about 35% two years after. We also observe evidence suggesting a reduction in the electricity cost per unit of output due to foreign acquisition, although the estimates are less precise. The magnitude of reduction is also consistently lower than that of fuel and lubricants, which might suggest that the expansion in production associated with foreign acquisition is largely driven by increased consumption of electricity.

We conjecture that expansion of acquired firms is fueled mostly by electricity because of the relatively lower electricity rates prevailing in Indonesia compared to its neighboring countries. This conjecture is consistent with the findings of other studies suggesting that the kind of technology a firm will use to reduce emissions (either through pollution abatement or increasing energy efficiency) may be effectively influenced by policies that directly affect factor prices (see, for example, Khanna and Zilberman, 2001; Harrison et al., 2015). Indonesia's lower industrial electricity tariffs might also explain why a lot of FDI inflows into the country during the sample period were directed to electricity-intensive industries (e.g., manufacture of machinery). The low electricity tariffs are due to the country's generous subsidies (Mourougane, 2010).

We also test whether the acquired plants' reallocation are due to possible energy subsidies provided by the government to foreign investors or multinationals. In other words, we check if acquired plants faced lower fuel and electricity prices after the acquisition and in subsequent years. We find no evidence in favor of this view (see Table G.1).

6.2 Are our findings driven purely by economies of scale?

As we have shown, acquired plants tend to expand their production significantly after the acquisition. However, larger output can be associated with less energy per unit of output if there are economies of scale in energy use. Therefore, it is possible that the improvement in energy efficiency in the acquired plants stems purely from expansion of output, and thus are not necessarily specific to foreign acquisition.²³

In order to address this concern, we examine the relationship between energy expenditure and output allowing for a different intercept and a different slope for the acquired plants.²⁴ More specifically, we estimate the following equation:

$$exp_{it} = \alpha + \beta_1(Post_t * Acquired_i) + \beta_2 y_{i,t-1} + \beta_3(Post_t * Acquired_i) * y_{i,t-1} + \gamma_i + \delta_t + \varepsilon_{it}$$
(5)

where exp_{it} is firm *i*'s energy expenditure (in logarithm) at year *t*, *Acquired* is foreign acquisition dummy as previously defined, *Post* is the post-acquisition period, *y* is real output (in logarithm) and γ_i and δ_t are plant and year fixed effects, respectively. We cluster standard error at the plant level.

We estimate equation 5 using the full as well as the matched sample. We then plot the estimated relationship for varying lagged output levels to get the combined marginal effect of the regressors at different lagged output levels, while eliminating common-to-firms and year-specific factors that may confound the results. We do so separately for acquired plants and for plants that remained in domestic hands. The idea is that if energy intensity is all attributed to the scale effect, then we can expect the trajectory of energy use as output grows to be parallel between the two groups. In other words, we want to test whether foreign acquisition affects energy use, controlling for the effect of economies of scale.

Panels (a) and (b) of Figure G.1 illustrate the results from estimating equation 5 on the unmatched and the matched sample, respectively.²⁵ In both panels, the lines are clearly not parallel between the two groups. In particular, the acquired plants have flatter relationship compared to that of domestic plants suggesting a smaller increase in energy use as the output expands. Even if we choose comparatively similar domestic plants, energy expenditure for acquired plants rises at a significantly slower rate (see Panel b).

We also find evidence to support the view that domestic plants undergo structural changes once they

²³Note, however, that even if the improvement in energy intensity were due to expansion of output, these economiesof-scale effect of foreign acquisitions would still lead to higher energy efficiency at the aggregate level. We will come back to this issue later in the paper.

 $^{^{24}}$ We also estimate a regression using contemporaneous output a regressor, taking note that this regression is subject to serious reverse causality issues. Notwithstanding, our results remain consistent with the baseline regression.

 $^{^{25}}$ For the detailed regression result, see Table G.2.

are acquired by foreign firms. In particular, we find that while acquired plants tend to increase both their capital stock and employment, they also tend to become more capital-intensive (see Table G.3). We do not observe strong evidence suggesting that acquired plants tend to invest more, although we see very strong and economically significant increase in purchases of machinery. The above observations, coupled with the fact that we see an indication of relatively higher reliance on electricity, suggests that our results cannot be just driven completely by the scale effect.

6.3 Are our findings driven purely by changes in the product mix?

It is possible that the improvement in energy efficiency observed in the aftermath of a foreign acquisition is due to changes in the mix of products produced by acquired plants. In other words, foreign owners could be increasing energy efficiency solely by shifting the output profile of the acquired plant towards less energy-intensive goods.

To shed light on this issue, we bring in new data that allow us to observe the value of production at the level of 5-digit ISIC products for each plant during the 1998-2008 period.²⁶ We use the new data to calculate the Hirschman-Herfindahl Index (HHI) of production concentration for each plant.²⁷ We then focus on plants with next to no change in the output mix (i.e., less than a 5% change in their HHI throughout the period considered) in the belief that such plants were unlikely to shift their production structure to cleaner products and examine whether we can still observe improvements in energy efficiency in these plants.

The new data also allow us to calculate the average energy intensity of each 5-digit ISIC product produced by Indonesian plants in years 2000-2008. We focus only on domestic plants, as we are interested in capturing energy intensity of the domestic production process. We choose not to use the years 1998 and 1999 because they were affected by the Asian crisis. In cases where no domestic plant produced a given product, we assign the average value for the 4-digit industry to the product.

We follow the same one-to-one propensity score matching as in our baseline analysis. However, our matching exercise is based just on plants with little change in the production structure (as defined above). We still match treated plants with control plants from the same industry-year cell. As shown in Table H.1, this process ensures that the treated and the control group are balanced in terms of output and energy efficiency in each of the two years prior to the ownership change. They are also balanced in terms of many characteristics not used in matching. In terms of energy intensity

²⁶The time period covered by the new data is different from the period covered by our baseline estimation (1983-2001) because we were unable to get access to product-level information in earlier years. Moreover, our baseline analysis finishes in 2001 because the post-2001 surveys no longer report me important energy inputs (such as, for instance, natural gas) as a separate category. This makes it impossible to estimate the total energy use in physical units and to convert it into CO_2 emissions.

²⁷The HHI is defined as the sum of squared product shares within each plant: $HHI_{it} = \sum_{k \in Kit} s_{kit}^2$, where s_k is the output share of product k produced by plant i in year t.

Log (Energy Ex	xpenditure/Output)		
	Acquisition Year	1 Year Later	2 Years Later
Post*Acquired	-0.548**	-0.528*	-0.442*
-	(0.276)	(0.285)	(0.254)
R-sq. (within)	0.036	0.033	0.035
Obs	222	222	222
	Acquisition Year	1 Year Later	2 Years Later
Post*Acquired	-1.854***	-1.727^{**}	-1.503**
-	(0.676)	(0.718)	(0.680)
Post*Acquired*log(Pre-acquisition predicted	0.023*	0.021	0.019
energy intensity)	(0.013)	(0.013)	(0.012)
R-sq. (within)	0.072	0.061	0.062
Obs	222	222	222
	Acquisition Year	1 Year Later	2 Years Later
Post*Acquired	-1.590^{**}	-1.465^{**}	-1.116
	(0.705)	(0.729)	(0.699)
Post*Acquired*log(Pre-acquisition predicted	0.019	0.018	0.014
energy intensity)	(0.013)	(0.013)	(0.012)
Log (Output)	-0.205	-0.179	-0.221
	(0.163)	(0.153)	(0.148)
	0.109	0.004	0 105
R-sq. (within)	0.103	0.084	0.100
	All years	All years	ll years
Post*Acquired	-0.078	-1.694^{**}	-1.516^{**}
	(0.258)	(0.670)	(0.665)
$Post*Acquired*log(Pre-acquisition\ predicted$		0.021^{*}	0.018
energy intensity)		(0.012)	(0.012)
Log (Output)			-0.119
			(0.087)
R-sq. (within)	0.025	0.048	0.062
Obs	444	444	444

Table 7: Robustness check on additional data: Plants with little change in
product mix.

Note: The table reflects the result of estimating equation 4 on the matched sample utilizing new data. Estimates are relative to pre-acquisition period. Each column in each panel is a separate regression for a particular outcome variable covering the two time periods. Heteroskedasticity-robust standard errors clustered at the plant level are in parentheses. *Post* dummy, its interaction with pre-acquisition energy intensity and plant fixed effects are included in all specifications. *, **, *** indicate statistical significance at 0.10, 0.05, and 0.01 level, respectively.

predicted by their output structure (defined as the product-specific energy intensity weighted by output shares), they appear to be slightly less energy efficient (with test p-values of 0.042 and 0.089 when the predicted energy efficiency is based on a continuous measure of product energy intensity and percentile of product energy intensity in the distribution for all products, respectively).

The results, presented in the top panel of Table 7, show a significant decline in energy intensity taking place after a foreign acquisition. The magnitudes are somewhat larger than those found in the baseline analysis. In the next panel of the same table, we allow for the impact of a foreign acquisition to differ based on the pre-acquisition energy intensity predicted from the plant production structure. This augmented model confirms the link between foreign acquisitions and improvements in energy efficiency. The pre-acquisition production structure does not appear to matter. In the third panel, we additionally control for contemporaneous changes in output aiming in this way to eliminate the possibility that scale effects are responsible for our findings. The results for the first two periods are robust but they cease to be statistically significant in the last period. As our sample in this exercise is based on a relatively small number of observations, in the lowest panel of the table we repeat all three specifications but pool together observations for all years.²⁸ The pooled regressions confirm our previous findings that foreign acquisitions are associated with a decline in energy intensity.

To summarize, the aim of this subsection was to shed light on the channels through which foreign ownership may lead better to energy performance. By focusing on plants with next to no changes to the output mix, we first eliminated the possibility that foreign ownership works solely through changes in the production structure. Then by controlling for the contemporaneous output we shut down the scale channel. The fact that we still found a positive impact of foreign acquisitions on energy efficiency is very suggestive of foreign acquisitions being associated with improvements to the production process taking place through the introduction of better technologies or better management.

In an unreported additional exercise, we checked whether foreign acquisitions in general resulted in shifting the output mix of acquired plants towards less energy-intensive products. We did not uncover any statistically significant patterns.

6.4 Pre-acquisition energy intensity

The evidence presented thus far suggests that foreign acquisitions lower energy consumption and emission levels per unit of output. Nonetheless, it is not obvious whether these effects are similar across all plants. It could be that the effect of a change in ownership on plant-level environmental performance is stronger for plants that were initially smaller and less energy efficient. In other words, the improvement in the production technique may be larger for plants that were initially

²⁸Note that in each exercise in the previous panels, half of all the observations (N) pertain to the pre-acquisition period. Thus in the pooled exercise the total number of observations is not 3N but rather (3N+N)/2 = 2N.

further from the technological frontier.

To test this hypothesis, we estimate equation 4 augmented with an interaction term between the foreign acquisition dummy and the plant's actual energy intensity level in the pre-acquisition year. We focus on the period one year after the acquisition where we expect the treatment to have the most significant effect. The results are presented in Table 8 and are also plotted in Figure 5. In the figure, the solid line corresponds to point estimates, while dashed lines denote 95% confidence intervals. We observe that the magnitude of the decline in the cost of energy per unit of output is larger at higher levels of pre-acquisition energy intensity. This is also true of energy use and emissions per unit of output. These results indicate that relatively more energy intensive (and perhaps less efficient and smaller) plants tend to benefit more in terms of reducing energy use and emission intensities from foreign acquisitions. This finding might explain why previous literature had mixed results on the effect of foreign acquisitions on plant-level environmental performance (see, for example, Eskeland and Harrison, 2003; Cole et al., 2008a).

Figure 5: Effect of acquisition on energy and emission intensities at varying pre-acquisition energy intensities.



The figure illustrates estimated combined coefficients of foreign acquisition dummy and its interaction with preacquisition energy intensity in equation 4 using the matched sample. The dashed lines correspond to the 95-percent confidence interval. The period focuses at one year after the acquisition (i.e., t + 1) and estimates are relative to the pre-acquisition period.

Table 8: Matched difference-in-differences estimates: Testing for non-linear effect of acquisition at varying pre-acquisition energy intensity.

	Log(Energy Expenditure/Output)	Log(Energy Use/Output)	$Log(CO_2 Emissions/Output)$
Post*Acquired	-1.759^{***} (0.359)	-1.738^{***} (0.382)	-1.643^{***} (0.358)
Post*Acquired*Pre-acq. Energy Intensity	-0.409*** (0.092)	-0.392*** (0.097)	-0.378*** (0.093)
R-sq. (within)	0.213	0.191	0.185
No. of Obs.	814	814	814
Threshold Pre-acq. Energy Intensity:threshold	-4.30	4.43	-4.35
(% share of treated plants)	60.48	68.57	64.29

Note: The table reflects the result of estimating equation 4 on the matched sample described in Section 4. The period is one year after the acquisition (i.e., t+1) and estimates are relative to pre-acquisition period. The dependent variables are as listed in each panel and expressed in logarithms. Each column in each panel is a separate regression for a particular outcome variable covering the two time periods Except for the dummy variables, all variables are in logarithm. Heteroskedasticity-robust standard errors clustered at the plant level are in parentheses. *Post* dummy, its interaction with pre-acquisition energy intensity and plant fixed effects are included in all specifications. *, **, *** indicate statistical significance at 0.10, 0.05, and 0.01 level, respectively.

7 Do Foreign Divestments Lead to Lower Energy Efficiency?

If foreign acquisitions lead to improvements in energy efficiency, it is not unreasonable to expect that foreign divestments (i.e., foreign affiliates being sold by their parents to local owners) would have the opposite effect. This would be consistent with the existing literature that finds that foreign divestments wipe out some of the benefits brought by foreign ownership. For instance, Javorcik and Poelhekke (2017) find that foreign divestments in Indonesia are associated with a drop in the total factor productivity and a decline in output, markups, as well as export and import intensities.

Javorcik and Poelhekke (2017) discuss the factors that can explain why divestments might take place. The first set of factors may results from shocks experienced by the parent company and its home country. The second set of factors relate to the whole network of subsidiaries belonging to the parent company and divestments may result from relative changes in growth rates, production costs, regulation and others in all countries where these subsidiaries operate. The third set of factors relate to the affiliate's characteristics and performance, which may be significant in terms of the level of uncertainties during the acquisition period or when coupled with certain events that influences the affiliate's profitability. Finally, divestments may occur as a result of shocks associated with buyer's purchasing behavior.

We consider the impact of losing a foreign parent on energy-related performance. We follow the same approach as before (see Section 4), but rather than focusing on foreign acquisitions we consider the cases of divestment. More specifically, we consider plants that had at least 20% of foreign equity and where foreign ownership dropped to less than 20% and remained below this threshold for at

least three years. We compare the performance of these divested plants to the performance of the plants that remained in foreign hands. This is different from the previous analysis where our control plants are those that remain domestic. We perform this exercise first using the OLS and then focusing on the matched sample.

7.1 OLS results

The results of our difference-in-differences estimation on the unmatched sample, which ignores the selection bias and controls only for 4-digit-ISIC-industry-year fixed effects, are presented in Panel (A) of Table 9. The sample includes all divested foreign affiliates and all affiliates remaining under foreign ownership throughout. We find that divested affiliates experience a large, persistent, and statistically significant drop in output. The output shrinkage is accompanied by a decline in energy consumption and CO_2 emissions. More importantly from the perspective of our study, we observe a persistent and statistically significant increase in energy cost per output and emission intensity.

In Panel (B), we additionally control for unobserved firm heterogeneity to take into account predivestment characteristics of divested affiliates. The effects of divestment on output, energy and emission levels and intensities are all consistent with the previous estimation procedure. However, the magnitude of effect are uniformly smaller in magnitude, which indicates that it is important to address selection bias in the analysis.

in Panel (C), we repeat the exercise in Panel (B) but drop cases that are not included in our matched sample. Doing so makes the estimates slightly larger than in Panel (B), while maintaining the statistical significance. Nonetheless, these estimates are much smaller than those from Panel (A) where unobserved firm heterogeneity is not taken into account.

		Panel (A)			Panel (B)			Panel (C)	
	Always fo	preign and dive	sted plants	Always for	eign and dives	sted plants	Always foreign	and matched d	livested plants
	t	t+1	$^{t+2}$	t	t+1	t+2	t	t+1	$^{t+2}$
Log(Output)	-1.248^{***}	-1.254^{***}	-1.248^{***}	-0.490***	-0.501***	-0.497***	-0.540^{***}	-0.544^{***}	-0.536***
Log(Energy Expenditure in Rps)	-1.001^{***}	-0.999***	-0.995***	-0.392***	-0.385***	-0.388***	-0.427***	-0.421^{***}	-0.426^{***}
Log(Energy Use in MBTUs)	-0.990***	-0.989***	-0.984^{***}	-0.367***	-0.362***	-0.365^{***}	-0.402^{***}	-0.396^{***}	-0.403^{***}
Log (CO ₂ Emissions)	-0.990***	-0.989***	-0.984***	-0.370***	-0.366***	-0.369***	-0.405***	-0.399***	-0.407^{***}
Log(Energy Expenditure/Output)	0.227^{***}	0.234^{***}	0.234^{***}	0.093^{***}	0.108^{***}	0.105^{***}	0.104^{***}	0.114^{***}	0.104^{***}
Log(Energy Use/Output)	0.238^{***}	0.244^{***}	0.245^{***}	0.118^{***}	0.131^{***}	0.127^{***}	0.130^{***}	0.139^{***}	0.127^{***}
Log (CO ₂ Emissions/Output)	0.238^{***}	0.244^{***}	0.244^{***}	0.115^{***}	0.127^{***}	0.123^{***}	0.127^{***}	0.136^{***}	0.123^{***}
Log(Energy Expenditure/Materials)	0.243^{***}	0.251^{***}	0.254^{***}	0.098^{***}	0.113^{***}	0.117^{***}	0.119^{***}	0.131^{***}	0.126^{***}
Log(Energy Use/Materials)	0.253^{***}	0.260^{***}	0.264^{***}	0.124^{***}	0.137^{***}	0.141^{***}	0.146^{***}	0.157^{***}	0.150^{***}
Log (CO ₂ Emissions/Materials)	0.253^{***}	0.260^{***}	0.264^{***}	0.123^{***}	0.135^{***}	0.139^{***}	0.145^{***}	0.156^{***}	0.149^{***}
Industry-year fixed effects	Yes	Yes	Yes	Yes	Yes	\mathbf{Yes}	\mathbf{Yes}	Yes	Yes
Plant fixed effects	N_{O}	N_{O}	N_{O}	Yes	Yes	\mathbf{Yes}	Yes	Yes	\mathbf{Yes}
Sample Size		22566-23926			22400 - 23807			21722 - 23039	
Motor The telle about the mountee	J:ff anon on	in differences	and and and the	10 40 40 00 000 000	I comele Th	a dow ow dow t	and the second second	a liated in cool	Land Pool

Table 9: Difference-in-differences on the unmatched sample: Divestments.

Note: The table shows the results of difference-in-differences analysis on the unmatched sample. The dependent variables are as listed in each panel. Each column in each panel is a separate regression for a particular outcome variable covering two time periods: the year relative to acquisition (as listed in each column) and a year before foreign-owned plants were acquired. Heteroskedasticity-robust standard errors clustered at the plant level are in parentheses. *, **, *** indicate statistical significance at 0.10, 0.05, and 0.01 level, respectively.

37

7.2 Results from Difference-in-Differences on matched sample

We now turn to our difference-in-differences analysis on the matched sample. Here, we are able to analyze 256 divested plants (out of 597 divestment cases) for the period 1983-2001 that had carefully selected control plants within the 4-digit-ISIC-industry-year cell.²⁹ We follow the matching procedure in Section 4, resulting in a well-balanced matched sample even in variables that were not used in the matching (see the last panel of Table I.1).

The results are summarized in Table 10. We find that divested plants experience a drop in output relative to the control group. Output declines by 27% in the year of ownership change and the decline persists in the two subsequent years. In other words, divested affiliates would have seen a much faster increase in output had they remained foreign owned.

The decline in output is accompanied by an increase in energy cost per unit of output, starting with a 34% rise in the year of divestment and persisting in the two subsequent years. We observe the same pattern for physical energy use (in MBTUs) and CO₂ emissions per unit of output. We also note that the effect of divestment, while pointing in the opposite direction, is quite similar in magnitude to the effect of acquisition on energy and emission intensities.

We also normalize the energy use and emission levels using material expenditure to test for the possible effect of reduced markups. The idea is that divested plants may rely less on exports, which consequently would reduce their markups (Javorcik and Poelhekke, 2017). If this is the case, then we should expect to see a larger effect when we normalize energy and emission levels with a variable that is less likely to be influenced by markups. This conjecture finds some support in the data. We find that divestment causes a plant's energy expenditure (and use in MBTUs) per unit of material expenditure to increase by 46%. This effect persists in two years after the year of divestment. The pattern also holds for CO_2 emission levels when normalized by expenditures on materials.

Overall, these findings are suggestive of the deterioration in energy efficiency that may be due to the loss of headquarter services and foreign managers after the ownership change. It is also possible that after foreign divestment the plant is unable to use the same production process, perhaps due to licensing requirements or other restrictions related to protection of intellectual property rights.

²⁹During our sample period, there were 1,174 plants that remained on foreign hands throughout.

	Divestment Year	1 Year Later	2 Years Later
Post*Divested R-sq. (within) No. of Obs.	-0.318*** (0.081) 0.030 1024	Log(Output) -0.397*** (0.092) 0.038 1024	-0.313*** (0.091) 0.035 1024
	Log (Enor	or Emonditure	(Output)
Post*Divested	0.296***	0.406***	0.290**
P so (within)	(0.099)	(0.108)	(0.121)
No. of Obs.	1022	1022	1022
	Log	Energy Use/Out	-put)
Post*Divested	0.296***	0.454***	0.258**
	(0.106)	(0.119)	(0.126)
R-sq. (within)	0.019	0.036	0.017
No. of Obs.	1022	1022	1022
	Log (Co	$O_2 \text{ Emissions}/O$	utput)
Post*Divested	0.289***	0.453***	0.249**
	(0.106)	(0.120)	(0.126)
R-sq. (within)	0.019	0.036	0.018
No. of Obs.	1022	1022	1022
	Log(Energ	y Expenditure/I	Materials)
Post*Divested	0.382***	0.429***	0.369***
	(0.111)	(0.122)	(0.136)
R-sq. (within)	0.030	0.040	0.029
No. of Obs.	1007	1003	1002
	Log (E	nergy Use/Mate	erials)
Post*Divested	0.381***	0.478***	0.338**
	(0.118)	(0.133)	(0.143)
R-sq. (within)	0.027	0.040	0.030
No. of Obs.	1007	1003	1002
	Log (CO	P_2 Emissions/Ma	aterials)
Post*Divested	0.375***	0.476***	0.327**
	(0.118)	(0.133)	(0.143)
R-sq. (within)	0.027	0.041	0.033
No. of Obs.	1007	1003	1002

Table 10: Difference-in-differences analysis on the matched sample: Divestments

Note: The table reflects the result of estimating equation 4 on the matched sample described in Section 4, but focusing on divestments. The dependent variables are as listed in each panel. Each column in each panel is a separate regression for a particular outcome variable covering two time periods: the year relative to divestment (as listed in each column) and a year before initially foreign-owned plants become domestic. Heteroskedasticity-robust standard errors clustered at the plant level are in parentheses. *Post* dummy and plant fixed effects are included in all specifications. *, **, *** indicate statistical significance at 0.10, 0.05, and 0.01 level, respectively.

8 Conclusions

This study contributes to literature by examining the effect of FDI on plant-level environmental performance. More specifically, it asks how foreign acquisitions influence plant-level energy and CO_2 emission intensities using data from the Census of Indonesian Manufacturing covering the period 1983-2001. Our analysis improves on the previous literature in three important respects. First, we focus on physical consumption of energy and carbon dioxide emissions, instead of relying on energy expenditure which is a less suitable proxy. Second, we examine the changes in ownership taking place within the same plant. This allows us to focus on changes in plant-level energy efficiency introduced by foreign ownership. Third, we also consider foreign divestments and demonstrates that they undo the positive effects of foreign acquisitions.

Our measure of pollution impact is far from perfect. The analysis is limited to global pollution (i.e., CO_2 emissions) and ignores the impact of FDI on local pollutants, which may be of greater interest when analyzing the impact of cross-country differences in local environmental regulations. Devising a close approximation of plant-level emissions of local pollutants is extremely challenging as it requires information on plant-level pollution abatement which is not available in our data set. Moreover, our analysis does not address the potential technological and management spillover effects of FDI on domestic plants or the possible outsourcing of the "dirty" part of the multinational to other sectors or other countries.

Despite these methodological imperfections, the data show remarkable plant-level operational changes associated with ownership changes. First, we see a positive and significant effect of a foreign acquisition on plant output, and consequently, its levels of energy use and CO_2 emissions. More interestingly from the perspective of the study is the second effect, namely, the improvement in the efficiency of using energy inputs due to FDI-induced innovations and investments. We see indications of falling energy use and emission intensities, which implies that each additional unit of output is produced with lower energy and CO_2 content. There is also an indication that foreign acquisitions are actually facilitating structural changes in production processes relating to energy use and emissions, suggesting that the improvement in energy use is not solely due to economies of scale or changes in the product mix. The opposite effects are observed when we analyze the effect of divestments on plant-level energy and CO_2 emission intensities.

At the aggregate level, we find that the entry of foreign affiliates is negatively related to industrywide average energy intensity. It is possible that, besides the direct effects, increased foreign participation in the domestic market may also have an indirect effect on other plants' energy consumption and emission patterns. More future research is, however, needed to examine the developments in energy and emission efficiency spillovers in the aftermath of foreign acquisitions and divestments. Our findings also provide a way to reconcile the conflicting results within the broad literature of foreign ownership and plant-level environmental performance. Our results indicate that the discrepancy between the findings of previous studies may be a consequence of failing to account for the initial energy intensity (or efficiency) levels of acquired plants. The impact of acquisition on more energy intensive domestic plants is found to be more significant and larger than on those that are already less energy intensive (and presumably more efficient) domestic plants. Moving forward, it would interesting to empirically investigate other dimensions influencing the heterogeneity of acquisition-induced effects on plant-level emissions.

The results of our study have broader implications, particularly in terms of promoting "green growth" in response to the threats of climate change and environmental degradation. More specifically, they suggest that FDI may serve as a channel for international transfer of environmentallyfriendly technologies and practices, thus directly contributing to environmental progress.

References

References

- Akagi, S., Yokelson, R. J., Wiedinmyer, C., Alvarado, M., Reid, J., Karl, T., Crounse, J., and Wennberg, P. (2011). Emission factors for open and domestic biomass burning for use in atmospheric models. Atmospheric Chemistry and Physics, 11(9):4039–4072.
- Albornoz, F., Cole, M. A., Elliott, R. J. R., and Ercolani, M. G. (2009). In search of environmental spillovers. World Economy, 32(1):136–163.
- Alfaro, L. and Chen, M. X. (2012). Surviving the global financial crisis: foreign ownership and establishment performance. *American Economic Journal: Economic Policy*, 4(3):30–55.
- Anderson, S. T. and Newell, R. G. (2004). Information programs for technology adoption: the case of energy-efficiency audits. *Resource and Energy Economics*, 26(1):27–50.
- Arnold, J. M. and Javorcik, B. S. (2009). Gifted kids or pushy parents? Foreign direct investment and plant productivity in Indonesia. *Journal of International Economics*, 79(1):42 53.
- Austin, P. C. (2011). An introduction to propensity score methods for reducing the effects of confounding in observational studies. *Multivariate behavioral research*, 46(3):399–424.
- Austin, P. C., Grootendorst, P., and Anderson, G. M. (2007). A comparison of the ability of different propensity score models to balance measured variables between treated and untreated subjects: a Monte Carlo study. *Statistics in Medicine*, 26(4):734–753.
- Azoulay, P., Graff Zivin, J. S., and Wang, J. (2010). Superstar extinction. The Quarterly Journal of Economics, 125(2):549–589.
- Barrows, G. and Ollivier, H. (2014). Does trade make firms cleaner? theory and evidence from indian manufacturing.
- Barrows, G. and Ollivier, H. (2018). Cleaner firms or cleaner products? how product mix shapes emission intensity from manufacturing. *Journal of Environmental Economics and Management*, 88:134 – 158.
- Beatty, T. K. and Tuttle, C. J. (2015). Expenditure response to Increases in in-kind transfers: Evidence from the Supplemental Nutrition Assistance Program. American Journal of Agricultural Economics, 97(2):390–404.
- Best, D., Mulyana, R., Jacobs, B., Iskandar, U. P., and Beck, B. (2011). Status of CCS development in Indonesia. *Energy Procedia*, 4:6152–6156.

- Bircan, C. (2019). Ownership structure and productivity of multinationals. Technical report.
- Blackman, A. and Wu, X. (1999). Foreign direct investment in China's power sector: trends, benefits and barriers. *Energy Policy*, 27(12):695 – 711.
- Blalock, G., Gertler, P. J., and Levine, D. I. (2008). Financial constraints on investment in an emerging market crisis. *Journal of Monetary Economics*, 55(3):568–591.
- Bloom, N., Eifert, B., Mahajan, A., McKenzie, D., and Roberts, J. (2013). Does management matter? evidence from india. *The Quarterly Journal of Economics*, 128(1):1–51.
- Bloom, N., Sadun, R., and Van Reenen, J. (2012). The organization of firms across countries. The Quarterly Journal of Economics, 127(4):1663–1705.
- Busso, M., DiNardo, J., and McCrary, J. (2014). New evidence on the finite sample properties of propensity score reweighting and matching estimators. *Review of Economics and Statistics*, 96(5):885–897.
- Byrne, R., de Coninck, H., and Sagar, A. (2014). Low-carbon innovation for industrial sectors in developing countries: Policy brief, january 2014. Technical report, Climate Strategies.
- Cai, X., Lu, Y., Wu, M., and Yu, L. (2016). Does environmental regulation drive away inbound foreign direct investment? Evidence from a quasi-natural experiment in China. *Journal of Development Economics*, 123:73–85.
- Chachavalpongpun, P. (2013). Global demand for palm oil drives FDI in Indonesian plantations and rapid land clearing. http://yaleglobal.yale.edu/content/politics-palm-oil.
- Chen, W. (2011). The effect of investor origin on firm performance: Domestic and foreign direct investment in the United States. *Journal of International Economics*, 83(2):219–228.
- Cole, M. A., Elliott, R. J., and Strobl, E. (2008a). The environmental performance of firms: The role of foreign ownership, training, and experience. *Ecological Economics*, 65(3):538 546.
- Cole, M. A., Elliott, R. J., and Wu, S. (2008b). activity and the environment in China: an industry-level analysis. *China Economic Review*, 19(3):393 408.
- Cole, M. A., Elliott, R. J., and Zhang, J. (2011). Growth, foreign direct investment, and the environment: evidence from Chinese cities. *Journal of Regional Science*, 51(1):121–138.
- Cole, M. A., Elliott, R. J., and Zhang, L. (2017). Foreign direct investment and the environment. Annual Review of Environment and Resources, 42:465–487.
- Dean, J. M., Lovely, M. E., and Wang, H. (2009). Are foreign investors attracted to weak environ-

mental regulations? Evaluating the evidence from China. Journal of Development Economics, 90(1):1 - 13.

- DeCanio, S. J. (1993). Barriers within firms to energy-efficient investments. *Energy policy*, 21(9):906–914.
- DeCanio, S. J. (1998). The efficiency paradox: bureaucratic and organizational barriers to profitable energy-saving investments. *Energy policy*, 26(5):441–454.
- DeCanio, S. J. and Watkins, W. E. (1998). Investment in energy efficiency: do the characteristics of firms matter? *The Review of Economics and Statistics*, 80(1):95–107.
- Dehejia, R. H. and Wahba, S. (2002). Propensity score-matching methods for nonexperimental causal studies. *The Review of Economics and Statistics*, 84(1):151–161.
- Eskeland, G. S. and Harrison, A. E. (2003). Moving to greener pastures? multinationals and the pollution haven hypothesis. *Journal of Development Economics*, 70(1):1 23.
- Forslid, R., Okubo, T., and Ullveit-Moe, K. (2018). Why are firms that export cleaner? international trade, abatement and environmental emissions. Technical report.
- Garcia, J., Sterner, T., and Afsah, S. (2007). Public disclosure of industrial pollution: the PROPER approach for Indonesia? *Environment and Development Economics*, 12:739–756.
- Gillingham, K., Newell, R. G., and Palmer, K. (2009). Energy efficiency economics and policy. Annual Review of Resource Economics, 1(1):597–620.
- Guadalupe, M., Kuzmina, O., and Thomas, C. (2012). Innovation and foreign ownership. American Economic Review, 102(7):3594–3627.
- Hanna, R. (2010). US environmental regulation and FDI: Evidence from a panel of US-based multinational firms. American Economic Journal: Applied Economics, 2(3):158–89.
- Harrison, A., Hyman, B., Martin, L., and Nataraj, S. (2015). When do firms go green? Comparing price incentives with command and control regulations in India. Working Paper 21763, National Bureau of Economic Research.
- Harrison, A. E., Martin, L. A., and Nataraj, S. (2012). Learning versus stealing: how important are market-share reallocations to India's productivity growth? The World Bank Economic Review, 27(2):202–228.
- Havranek, T. and Irsova, Z. (2011). Estimating vertical spillovers from fdi: Why results vary and what the true effect is. *Journal of International Economics*, 85(2):234–244.

- Hirano, K. and Imbens, G. W. (2001). Estimation of causal effects using propensity score weighting: An application to data on right heart catheterization. *Health Services and Outcomes Research Methodology*, 2(3-4):259–278.
- Iacus, S. M., King, G., and Porro, G. (2009). CEM: software for coarsened exact matching. Journal of Statistical Software, 30(9):1–27.
- Iacus, S. M., King, G., and Porro, G. (2011). Multivariate matching methods that are monotonic imbalance bounding. *Journal of the American Statistical Association*, 106(493):345–361.
- Javorcik, B. (2004). Does foreign direct investment increase the productivity of domestic firms? In search of spillovers through backward linkages. *American Economic Review*, 94(3):605–627.
- Javorcik, B. and Poelhekke, S. (2017). Former foreign affiliates: Cast out and outperformed? Journal of the European Economic Association, 15(3):501–539.
- Javorcik, B. and Wei, S.-J. (2004). Pollution, corruption and the location of foreign direct investment: Dirty secret or popular myth? *Contributions to Economic Analysis and Policy*, 3(2).
- Javorcik, B. S. and Li, Y. (2013). Do the biggest aisles serve a brighter future? Global retail chains and their implications for Romania. *Journal of International Economics*, 90(2):348–363.
- Kellenberg, D. K. (2009). An empirical investigation of the pollution haven effect with strategic environment and trade policy. *Journal of International Economics*, 78(2):242 255.
- Khanna, M. and Zilberman, D. (2001). Adoption of energy efficient technologies and carbon abatement: the electricity generating sector in India. *Energy Economics*, 23(6):637–658.
- Leuven, E. and Sianesi, B. (2012). Psmatch2: Stata module to perform full mahalanobis and propensity score matching, common support graphing, and covariate imbalance testing. *Statistical Software Components*.
- Millimet, D. and Roy, J. (2015). Empirical tests of the pollution haven hypothesis when environmental regulation is endogenous. *Journal of Applied Econometrics*, 31(4):652–677.
- Mourougane, A. (2010). Phasing out energy subsidies in Indonesia. OECD Economics Department Working Papers, 808.
- Olley, G. S. and Pakes, A. (1996). The Dynamics of Productivity in the Telecommunications Equipment Industry. *Econometrica*, 64(6):1263–1297.
- Pargal, S. and Wheeler, D. (1996). Informal regulation of industrial pollution in developing countries: evidence from Indonesia. *Journal of Political Economy*, 104(6):1314–1327.

- Pavcnik, N. (2002). Trade liberalization, exit, and productivity improvements: Evidence from Chilean plants. *The Review of Economic Studies*, 69(1):245–276.
- Rondinelli, D. A. and Berry, M. A. (2000). Environmental citizenship in multinational corporations: social responsibility and sustainable development. *European Management Journal*, 18(1):70–84.
- Rosenbaum, P. R. and Rubin, D. B. (1983). The central role of the propensity score in observational studies for causal effects. *Biometrika*, 70(1):41–55.
- Rosenbaum, P. R. and Rubin, D. B. (1985). Constructing a control group using multivariate matched sampling methods that incorporate the propensity score. *The American Statistician*, 39(1):33–38.
- Singh, J. and Agrawal, A. (2011). Recruiting for ideas: how firms exploit the prior inventions of new hires. *Management Science*, 57(1):129–150.
- Wang, J., Azoulay, P., and Zivin, J. G. (2010). Superstar extinction. Quarterly Journal of Economics, 125(2):549–589.
- Wang, J. and Wang, X. (2015). Benefits of foreign ownership: Evidence from foreign direct investment in China. Journal of International Economics, 97(2):325–338.
- Zarsky, L. (1999). Havens, halos and spaghetti: untangling the evidence about foreign direct investment and the environment. Foreign Direct Investment and the Environment, pages 47–74.

List of Appendices (for Online Publication)

Distribution of acquisition cases, matched and unmatched	48
Measuring Energy Use and Emissions, Conversion Factors	50
Tables and Figures for Robustness Checks	52
Coarsened exact matching	56
Inverse Probability of Treatment Weighting (IPTW)	59
Extended data: 1983-2008	62
Energy Prices, Scale Economies and Other Outcomes	65
Analysis of Channels: Additional Table	68
Divestment: Additional Table	69

Appendix A Distribution of acquisition cases, matched and unmatched



Figure A.1: Distribution of acquired plants, by sector.

Note: Industry classification is based on International Standard Industry Classification (ISIC) Rev2 2-digit level.

Source: Indonesian Census of Manufacturing



Figure A.2: Distribution of acquisition cases, by year.

Source: Indonesian Census of Manufacturing

Figure A.3: Distribution of acquisition cases, matched vs. unmatched, by industry.



Source: Indonesian Census of Manufacturing

Appendix B Measuring Energy Use and Emissions, Conversion Factors

	Obs.	R.sq.	Adj. R-sq.	RMSE
Fuel and Lubricant (Total)				
Gasoline	140,744	0.993	0.993	0.144
Diesel	198,309	0.992	0.992	0.208
Diesel Oil	30,279	0.966	0.966	0.564
Kerosene	104,509	0.696	0.695	1.164
Lubricant	213,705	0.944	0.944	0.520
Bunker Oil	2,049	0.999	0.999	0.102
Coal	$1,\!370$	0.975	0.973	0.640
Coke	2,912	0.942	0.940	0.633
Public Gas	9,598	0.931	0.930	0.812
LPG	11,906	0.993	0.993	0.175
Firewood	9,089	0.990	0.990	0.218
Charcoal	1,460	0.988	0.987	0.236
Fuel and Lubricant (Elec. Generation)				
Gasoline	$1,\!435$	0.983	0.982	0.303
Diesel	$37,\!683$	0.987	0.987	0.285
Diesel Oil	2,538	0.966	0.965	0.580
Kerosene	317	0.981	0.974	0.449
Lubricant	$22,\!606$	0.972	0.972	0.357
Bunker Oil	179	0.996	0.995	0.258
Coal	29	0.997	0.988	0.401
Coke	6	1.000		-
Public Gas	93	0.991	0.983	0.402
LPG	110	0.985	0.969	0.488
Firewood	44	0.983	0.945	0.760
Charcoal	16	1.000		-
Electricity Use				
Sold	702	0.978	0.974	0.489
PLN	$216,\!193$	0.979	0.979	0.349
Non-PLN	5,022	0.974	0.974	0.465

Table B.1: Goodness-of-Fit in estimating plant-level energy use (in physical units).

Note: The table reports the goodness-of-fit in estimating equation 1. Each variable is expressed in log. LPG denotes liquefied petroleum gas. PLN refers to amount of electricity bought from Indonesia's state-owned power company *Perusahaan Listrik Negar*, while non-PLN refers to those that are bought from independent power producers. Source: Indonesian Census of Manufacturing

Conversion to Energy (in MBTUs)	
Gasoline	Silverman, D. (Univ. of California, Irvine)
Diesel	US Energy Information Administration
Fuel Oil/Bunker Oil	US Energy Information Administration
Kerosene	US Energy Information Administration
Lubricants	US Energy Information Administration
Coal	US Environmental Protection Agency
Coke	US Energy Information Administration
Public Gas	US Bureau of Mines
Liquefied Petroleum Gas	US Environmental Protection Agency
Firewood	Silverman, D. (Univ. of California, Irvine)
Charcoal	Oak Ridge National Laboratory
Electricity	US Energy Information Administration
Conversion to Carbon Dioxide (in Kg/C)	
Gasoline	US Energy Information Administration
Diesel	US Environmental Protection Agency
Fuel Oil/Bunker Oil	US Environmental Protection Agency
Kerosene	US Environmental Protection Agency
Lubricants	US Energy Information Administration
Coal	US Energy Information Administration
Coke	US Energy Information Administration
Public Gas	US Energy Information Administration
Liquefied Petroleum Gas	US Energy Information Administration
Firewood	Partnership for Policy Integrity
Charcoal	Akagi et al. (2011)
Electricity	US Environmental Protection Agency

 Table B.2: Sources of Conversion Factors

Appendix C Tables and Figures for Robustness Checks



Figure C.1: Trajectories of output, energy expenditure and energy intensity: Acquired vs. domestic plants, excluding post-1997 crisis.

The figure illustrates the average value of each variable of interest in a given time period for the treated (acquired) and control (domestic) group. The horizontal axes indicate the year relative to the period t where treated plants are acquired.

	A	- T 7 - T -	0.17 T .			
	Acquisition Year	1 Year Later	2 Years Later			
		Log(Output)				
Post*Acquired	0.793^{***}	0.777^{***}	0.798^{***}			
	(0.125)	(0.134)	(0.156)			
R-sq. (within)	0.236	0.281	0.291			
No. of Obs.	714	654	614			
	Log (Ene	rgy expenditure	in Rps)			
Post*Acquired	0.519***	0.647***	0.492***			
	(0.133)	(0.152)	(0.184)			
R-sq. (within)	0.134	0.174	0.136			
No. of Obs.	714	654	613			
	Log (E	nergy use in MI	BTUs)			
Post*Acquired	0.486***	0.597***	0.474***			
-	(0.135)	(0.156)	(0.182)			
R-sq. (within)	0.130	0.155	0.147			
No. of Obs.	714	654	613			
	L_{OG} (CO ₂ emissions in kg CO ₂)					
Post*Acquired	0.494***	0.601***	0.473**			
1	(0.136)	(0.157)	(0.184)			
R-sq. (within)	0.142	0.167	0.158			
No. of Obs.	714	654	613			
	Log (Ener	gy expenditure.	(Output)			
Post*Acquired	-0.273**	-0.130	-0.310*			
1	(0.131)	(0.130)	(0.160)			
R-sq. (within)	0.019	0.012	0.031			
No. of Obs.	714	654	613			
	Log (Energy use/Out	mut)			
Post*Acquired	-0.307**	-0 180	-0.328**			
1 obti Hoquinou	(0.135)	(0.138)	(0.166)			
B-sq. (within)	0.020	0.015	0.026			
No. of Obs.	714	654	613			
	Log (O	O. omissions /O	utput)			
Post*Acquired	LOg (U _0 200**	-0 176	-0 329**			
1 050 Mequirea	(0.133)	(0.137)	(0.164)			
B-sa (within)	0.017	0.011	0.022			
No. of Obs	714	654	613			
110. 01 005.	114	004	010			

 Table C.1: Matched difference-in-differences estimates: Excluding post-1997 financial crisis period.

Note: The table reflects the result of estimating equation 4 on the matched sample described in Section 4 but drops years after 1997. The dependent variables are as listed in each panel. Each column in each panel is a separate regression for a particular outcome variable covering two time periods: the year relative to acquisition (as listed in each column) and a year before foreign-owned plants were acquired. Heteroskedasticity-robust standard errors clustered at the plant level are in parentheses. *Post* dummy and plant fixed effects are included in all specifications. *, **, *** indicate statistical significance at 0.10, 0.05, and 0.01 level, respectively.

$\begin{array}{c c c c c c c c c c c c c c c c c c c $		Acquisition Year	1 Year Later	2 Years Later	3 Years Later	4 Years Later	5 Years Later	
Post*Acquired 0.728*** 0.839*** 0.813*** 1.033*** 1.104*** 1.118*** (0.135) (0.139) (0.150) (0.170) (0.180) (0.192) R-sq. (within) 0.247 0.316 0.281 0.299 0.292 0.278 No. of Obs. 462 462 462 462 462 462 Log (Energy Expenditure in Rps)	L og(Output)							
R-sq. (within) 0.247 0.316 0.281 0.299 0.292 0.278 No. of Obs. 462 462 462 462 462 462	Post*Acquired	0 728***	0 839***	0.813***	1 033***	1 104***	1 118***	
R-sq. (within) 0.247 0.316 0.281 0.299 0.292 0.278 No. of Obs. 462 462 462 462 462 462	rost nequirea	(0.135)	(0.139)	(0.150)	(0.170)	(0.180)	(0.192)	
No. of Obs. 462 462 462 462 462 462 Log (Energy Expenditure in Rps)	R-sq. (within)	0.247	0.316	0.281	0.299	0.292	0.278	
Log (Energy Expenditure in Rps)	No. of Obs.	462	462	462	462	462	462	
Log (Energy Expenditure in typs)			-	og (Frorgy Fyr	onditure in Pre			
Post*Acquired 0.430*** 0.502*** 0.500*** 0.306 0.420** 0.647***	Post*Acquired	0 /30***	0 593***	0 500***	0.306	0 /20**	0 647***	
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	i ost Acquirea	(0.152)	(0.159)	(0.183)	(0.198)	(0.186)	(0.185)	
$R_{-sq} (within) = 0.124 = 0.184 = 0.141 = 0.138 = 0.206 = 0.221$	R-sa (within)	0.194	0.184	0.141	0.138	0.206	0.221	
No of Obs 454 454 454 454 454 454 454	No. of Obs	454	454	454	454	454	454	
	110. 01 0.05.	101	101			101	101	
Log (Energy Use in MBTUs)		0.007**	0 5 40***	Log (Energy U	se in MBTUs)	0.010*	0 500***	
Post*Acquired 0.337^{++} 0.542^{+++} 0.429^{++} 0.227 0.318^{+-} 0.598^{+++} (0.151) (0.162) (0.162) (0.162) (0.162) (0.162) (0.162)	Post*Acquired	0.337**	0.542^{***}	0.429^{**}	0.227	0.318^{*}	0.598^{***}	
(0.151) (0.162) (0.182) (0.200) (0.187) (0.193)	\mathbf{D} ((11)	(0.151)	(0.162)	(0.182)	(0.200)	(0.187)	(0.193)	
R-sq. (within) 0.112 0.166 0.149 0.153 0.208 0.200	R-sq. (within)	0.112	0.166	0.149	0.153	0.208	0.200	
No. of Obs. 454 454 454 454 454 454	No. of Obs.	454	454	454	454	454	454	
$Log (CO_2 \text{ Emissions in } kg CO_2)$								
Post*Acquired 0.314** 0.524*** 0.406** 0.192 0.311 0.589***	Post*Acquired	0.314**	0.524^{***}	0.406**	0.192	0.311	0.589^{***}	
$(0.154) \qquad (0.164) \qquad (0.184) \qquad (0.203) \qquad (0.191) \qquad (0.194)$		(0.154)	(0.164)	(0.184)	(0.203)	(0.191)	(0.194)	
R-sq. (within) 0.118 0.171 0.154 0.161 0.219 0.212	R-sq. (within)	0.118	0.171	0.154	0.161	0.219	0.212	
No. of Obs. 454 454 454 454 454 454	No. of Obs.	454	454	454	454	454	454	
Log (Energy Expenditure/Output)			Ι	og (Energy Exp	enditure/Output)		
Post*Acquired -0.308** -0.272** -0.345** -0.718*** -0.675*** -0.489***	Post*Acquired	-0.308**	-0.272**	-0.345**	-0.718***	-0.675***	-0.489***	
(0.134) (0.126) (0.161) (0.157) (0.155) (0.155)	_	(0.134)	(0.126)	(0.161)	(0.157)	(0.155)	(0.155)	
R-sq. (within) 0.025 0.024 0.019 0.087 0.084 0.037	R-sq. (within)	0.025	0.024	0.019	0.087	0.084	0.037	
No. of Obs. 454 454 454 454 454 454	No. of Obs.	454	454	454	454	454	454	
Log (Energy Use/Output)				Log (Energy	Use/Output)			
Post*Acquired -0.401*** -0.323** -0.416** -0.797*** -0.777*** -0.537***	Post*Acquired	-0.401***	-0.323**	-0.416**	-0.797***	-0.777***	-0.537***	
(0.136) (0.135) (0.164) (0.168) (0.164) (0.164)	*	(0.136)	(0.135)	(0.164)	(0.168)	(0.164)	(0.164)	
R-sq. (within) 0.041 0.030 0.025 0.091 0.099 0.040	R-sq. (within)	0.041	0.030	0.025	0.091	0.099	0.040	
No. of Obs. 454 454 454 454 454 454	No. of Obs.	454	454	454	454	454	454	
Log (CO ₂ Emissions/Output)				Log (CO ₂ Emi	ssions/Output)			
Post*Acquired -0.424*** -0.341** -0.439*** -0.832*** -0.784*** -0.547***	Post*Acquired	-0.424***	-0.341**	-0.439***	-0.832***	-0.784***	-0.547***	
(0.139) (0.136) (0.166) (0.172) (0.168) (0.165)	1 oot mequiled	(0.139)	(0.136)	(0.166)	(0.172)	(0.168)	(0.165)	
R-sq. (within) 0.039 0.028 0.026 0.095 0.101 0.042	R-sq. (within)	0.039	0.028	0.026	0.095	0.101	0.042	
No. of Obs. 454 454 454 454 454 454 454	No. of Obs.	454	454	454	454	454	454	

Table C.2: Constant sample over a longer time horizon.

Note: The table reflects the result of estimating equation 4 on the matched sample described in Section 4 but ensures that the included plants have non-missing observations in each time period. The dependent variables are as listed in each panel. Each column in each panel is a separate regression for a particular outcome variable covering two time periods: the year relative to acquisition (as listed in each column) and a year before foreign-owned plants were acquired. Heteroskedasticity-robust standard errors clustered at the plant level are in parentheses. *Post* dummy and plant fixed effects are included in all specifications. *, **, *** indicate statistical significance at 0.10, 0.05, and 0.01 level, respectively.

	Acquisition Year	1 Year Later	2 Years Later	3 Years Later	4 Years Later	5 Years Later	
Log(Output)							
Post*Acquired	0.707^{***}	0.827^{***}	0.803***	1.039***	1.126^{***}	1.090^{***}	
	(0.134)	(0.136)	(0.149)	(0.170)	(0.179)	(0.192)	
R-sq. (within)	0.242	0.315	0.276	0.302	0.295	0.267	
No. of Obs.	454	454	454	454	454	454	
]	Log (Energy Exp	enditure in Rps)			
Post*Acquired	0.442^{***}	0.589^{***}	0.547^{***}	0.363*	0.495^{***}	0.714^{***}	
	(0.154)	(0.159)	(0.186)	(0.201)	(0.188)	(0.189)	
R-sq. (within)	0.122	0.175	0.134	0.122	0.193	0.203	
No. of Obs.	446	446	446	446	446	446	
			Log (Energy U	se in MBTUs)			
Post*Acquired	0.354^{**}	0.569^{***}	0.467^{**}	0.268	0.392^{**}	0.663^{***}	
	(0.153)	(0.163)	(0.184)	(0.202)	(0.189)	(0.198)	
R-sq. (within)	0.109	0.163	0.141	0.136	0.195	0.183	
No. of Obs.	446	446	446	446	446	446	
	Log (CO2 Emissions in kg CO2)						
Post*Acquired	0.328^{**}	0.551^{***}	0.443**	0.232	0.381^{**}	0.654^{***}	
-	(0.156)	(0.166)	(0.187)	(0.206)	(0.193)	(0.198)	
R-sq. (within)	0.115	0.167	0.146	0.144	0.206	0.195	
No. of Obs.	446	446	446	446	446	446	
		Ι	og (Energy Exp	enditure/Output)		
Post*Acquired	-0.275**	-0.263**	-0.287^{*}	-0.667***	-0.624^{***}	-0.394**	
	(0.134)	(0.126)	(0.162)	(0.159)	(0.154)	(0.155)	
R-sq. (within)	0.020	0.024	0.014	0.077	0.072	0.023	
No. of Obs.	446	446	446	446	446	446	
			Log (Energy	Use/Output)			
Post*Acquired	-0.363***	-0.284^{**}	-0.368**	-0.763^{***}	-0.727^{***}	-0.444***	
	(0.136)	(0.133)	(0.165)	(0.169)	(0.163)	(0.165)	
R-sq. (within)	0.034	0.025	0.019	0.085	0.088	0.026	
No. of Obs.	446	446	446	446	446	446	
			Log (CO2 Emi	ssions/Output)			
Post*Acquired	-0.389***	-0.302**	-0.392^{**}	-0.798^{***}	-0.738***	-0.454^{***}	
	(0.139)	(0.134)	(0.167)	(0.173)	(0.167)	(0.166)	
R-sq. (within)	0.033	0.023	0.020	0.087	0.091	0.027	
No. of Obs.	446	446	446	446	446	446	

Table C.3: Constant sample over a longer time horizon (matching outside
the same county).

Note: The table reflects the result of estimating equation 4 on the matched sample (not belonging to the same province or *kabupaten*) described in Section 4 but ensures that the included plants have non-missing observations in each time period. The dependent variables are as listed in each panel. Each column in each panel is a separate regression for a particular outcome variable covering two time periods: the year relative to acquisition (as listed in each column) and a year before foreign-owned plants were acquired. Heteroskedasticity-robust standard errors clustered at the plant level are in parentheses. *Post* dummy and plant fixed effects are included in all specifications. *, **, *** indicate statistical significance at 0.10, 0.05, and 0.01 level, respectively.

Appendix D Coarsened exact matching

Coarsened exact matching (CEM) procedure (Iacus et al., 2009, 2011), unlike propensity score matching that relies on estimating a scalar (i.e., the propensity score), is a nonparametric method used to identify a control group for the treated observations. In our context, this is helpful because we do not observe the pollution abatement technology each plant has, which we think could be a factor that can help predict the probability of a plant being acquired. While it has some advantages over propensity score matching (e.g. balance on selected covariates *ex ante*), it suffers from the "curse of dimensionality" where the proportion of matched observations decreases rapidly with the number of strata (Azoulay et al., 2010).

The first step in implementing CEM is to identify a set of covariates on which we can balance the treated (i.e. acquired plants) and the control (i.e. domestic plants) observations. In our context, we used the same set of covariates that we used in the propensity score matching in order to maintain comparability. Similar to our baseline matching procedure, we implement CEM in each year-industry cell, thus creating 1,415 bins. In each bin, we coarsen the joint distribution of the selected covariates, resulting in about 200 strata in each bin on the average (ranging from 3 to 604 strata). Within each stratum, we identify a control plant for each of the treated plants. If there are multiple choices, ties are broken randomly. In this procedure, we are to identify a control plant for 264 acquired plants.

The first 4 columns of Table D.1 show that our matched treated and control plants are well balanced in the set of covariates that we used in the matching procedure. Similar to our baseline matching procedure, we also evaluate CEM on potential "selection on observables" bias by looking at mean differences between treated and control plants on a covariates that were not used in matching. In all of these covariates, we find no statistically significant mean differences. Meanwhile, Table D.2 shows that our baseline estimate is consistent even if we use a different matching procedure.

	CEM			PSM (no same county)			IPTW	
Variables		(N=418)			(N=440)		(N=1)	43,216)
	Treated	Control	p-value	Treated	Control	p-value	F-Stat	p-value
Used in matching								
$Log (Output)_{t-1}$	9.03	9.03	0.99	9.86	9.86	0.90	4.43	0.04
$Log (Energy Expenditure/Output)_{t-1}$	-3.71	-3.71	0.99	-3.82	-3.82	0.65	0.80	0.37
$Log (Output)_{t-2}$	9.59	9.62	0.90	9.86	9.86	0.90	5.91	0.02
Log (Energy Expenditure/Output) $_{t-2}$	-3.77	-3.79	0.92	-3.82	-3.82	0.65	0.02	0.89
Not used in matching								
$Log (Energy Expenditure)_{t-1}$	5.33	5.33	0.99	6.03	6.03	0.84	6.23	$0.01 \ 9$
$Log (Energy Use)_{t-1}$	8.25	8.26	0.95	8.99	8.99	0.77	5.42	0.01
$Log (CO2 Emission)_{t-1}$	12.65	12.66	0.94	13.36	13.36	0.78	0.02	0.02
$Log (Employment)_{t-1}$	4.85	4.72	0.26	5.26	5.26	0.40	4.59	0.00
Exporter $Dummy_{t-1}$	0.15	0.18	0.30	0.18	0.18	0.80	14.42	0.03
Share of Imported $Materials_{t-1}$	0.20	0.18	0.56	0.19	0.19	0.05	12.14	0.00
Share of Skilled Workers $_{t-1}$	0.19	0.20	0.63	0.21	0.21	0.25	17.91	0.00
$Log(Investment in Machinery)_{t-1}$	7.15	6.93	0.43	7.86	7.86	0.20	0.61	0.00
$Log(Energy Use/Output)_{t-1}$	-0.80	-0.79	0.93	-0.87	-0.87	0.56	0.22	0.43
Δ Log (Energy Expenditure) _{t-1}	0.14	0.08	0.19	0.15	0.15	0.55	0.03	0.90
Δ Log (Energy Use) _{t-1}	0.17	0.09	0.19	0.18	0.18	0.72	0.00	0.87
Δ Log (CO2 Emissions) _{t-1}	0.17	0.09	0.22	0.18	0.18	0.70	10.36	0.97
$Log(CO2 \text{ Emissions/Output})_{t-1}$	3.61	3.62	0.92	3.51	3.51	0.57	7.45	0.64
$Log(Energy Exp./Materials)_{t-1}$	-2.80	-2.81	0.93	-3.03	-3.03	0.32	0.39	0.01
Δ Log (Output) _{t-1}	0.13	0.05	0.14	0.14	0.14	0.86	2.05	0.53
Δ Log (Energy Expenditure/Output) _{t-1}	0.02	0.03	0.81	0.02	0.02	0.60	1.93	0.15
Δ Log(Energy Use/Output) _{t-1}	0.05	0.04	0.87	0.04	0.04	0.81	1.48	0.16
Δ Log(CO2 Emissions/Output) _{t-1}	0.04	0.04	0.96	0.04	0.04	0.78	1.66	0.22
Δ Log(Energy Exp./Materials) _{t-1}	0.08	-0.02	0.08	0.04	0.04	0.81	0.00	0.20

Table D.1: Balancing hypothesis under various matching procedures.

Note: CEM - coarsened exact matching; PSM - propensity score matching; IPW - inverse probability weighting procedure. The variables used in each matching procedure include lagged output, energy expenditure, energy use (in MBTUs) and CO_2 emission and the log difference on the last three variables.

	Acquisition Year	1 Year Later	2 Years Later		
Post*Acquired	$1.392^{***} \\ (0.141) \\ 0.350$	Log(Output) 1.499*** (0.144) 0.383	$1.530^{***} \\ (0.148) \\ 0.302$		
No. of Obs.	876	876	876		
	Log (Ene	rgy Expenditure	in Rps)		
Post*Acquired	1.012***	1.189***	1.159***		
1	(0.140)	(0.149)	(0.158)		
R-sq. (within)	0.221	0.248	0.253		
No. of Obs.	871	868	868		
	Log(Ei	nergy Use in MI	BTUs)		
Post*Acquired	0.992***	1.149***	1.085^{***}		
	(0.145)	(0.157)	(0.165)		
R-sq. (within)	0.209	0.232	0.229		
No. of Obs.	871	868	868		
	$Log (CO_2 \text{ Emissions})$				
Post*Acquired	1.000***	1.158***	1.099***		
	(0.145)	(0.156)	(0.165)		
R-sq. (within)	0.214	0.240	0.234		
No. of Obs.	871	868	868		
	Log (Ener	gy Expenditure	/Output)		
Post*Acquired	-0.372***	-0.297**	-0.382^{***}		
	(0.113)	(0.121)	(0.123)		
R-sq. (within)	0.059	0.054	0.048		
No. of Obs.	871	868	868		
	Log (Energy Use/Out	tput)		
Post*Acquired	-0.392***	-0.336**	-0.456***		
	(0.119)	(0.133)	(0.134)		
R-sq. (within)	0.053	0.042	0.048		
No. of Obs.	871	868	868		
	Log (Ce	$O_2 \text{ Emissions}/O$	utput)		
Post*Acquired	-0.383***	-0.327**	-0.443***		
	(0.119)	(0.131)	(0.133)		
R-sq. (within)	0.051	0.039	0.045		
No. of Obs.	871	868	868		

Table D.2: Difference-in-differences estimates: Coarsened exact matching.

Note: The table reflects the result of estimating equation 4 on the matched sample using CEM. The dependent variables are as listed in each panel. Each column in each panel is a separate regression for a particular outcome variable covering two time periods: the year relative to acquisition (as listed in each column) and a year before foreign-owned plants were acquired. Heteroskedasticity-robust standard errors clustered at the plant level are in parentheses. *Post* dummy and plant fixed effects are included in all specifications. *, **, *** indicate statistical significance at 0.10, 0.05, and 0.01 level, respectively.

Appendix E Inverse Probability of Treatment Weighting (IPTW)

As mentioned in the main text, one major drawback of the one-to-one propensity score matching is that we are limiting our conclusion to a very few matched observations. This may post an issue on the validity of our estimate if the majority of the dropped plants with acquisition cases are systematically different in our outcome variables but are marginally different in the set of covariates that we chose in the matching procedure. Moreover, there are also studies suggesting that the finite sample properties of one-to-one propensity score matching are inferior to other matching techniques (see, for example, Guadalupe et al., 2012; Busso et al., 2014).

To address this concern, we employ inverse probability of treatment weighting to identify the appropriate controls for each acquired plant. In particular, we transform the same propensity score estimated from the set of covariates in our baseline matching procedure into weights, following (Guadalupe et al., 2012). We weight each treated firm by $(1/\hat{p})$ and each control firm by $(1/(1-\hat{p}))$ where \hat{p} is our estimated propensity score from the baseline matching procedure. We apply these weights for samples which includes acquired plants that we observe 2 and 5 years after acquisition case to test for the comparability of the estimated average effect effect (ATE) with our period-by-period baseline estimates.

In contrast to our baseline procedure and CEM, we find statistically significant mean difference between treated and control plants on some observable covariates (see Table D.1). However, we find balanced sample in the trends of our variables of interests, which is a key assumption in performing difference-in-differences. The results from performing this alternative matching procedure are fairly consistent with our baseline estimates, suggesting that our baseline results are robust to different matching procedure (see Table E.1).

	Acquisition Year	1 Year Later	2 Years Later			
		Log(Output)				
Post*Acquired	1.659^{***}	1.763^{***}	1.906^{***}			
	(0.184)	(0.218)	(0.221)			
R-sq. (within)	0.803	0.807	0.809			
No. of Obs.	138750	138750	138750			
	Log (Ene	rgy Expenditure	e in Rps)			
Post*Acquired	1.353***	1.421^{***}	1.399^{***}			
-	(0.176)	(0.199)	(0.220)			
R-sq. (within)	0.806	0.802	0.800			
No. of Obs.	138011	138009	138008			
	Log(Er	nergy Use in MH	BTUs)			
Post*Acquired	1.284***	1.337^{***}	1.324^{***}			
*	(0.172)	(0.192)	(0.212)			
R-sq. (within)	0.805	0.798	0.795			
No. of Obs.	138011	138009	138008			
	Log (CO ₂ Emissions)					
Post*Acquired	1.359***	1.400***	1.395***			
1	(0.170)	(0.192)	(0.211)			
R-sq. (within)	0.806	0.799	0.794			
No. of Obs.	138010	138008	138007			
	Log (Ener	gy Expenditure	/Output)			
Post*Acquired	-0 324***	-0 358***	-0 516***			
i ost riequired	(0.120)	(0.133)	(0.159)			
B-sq (within)	0.640	0.649	0.638			
No. of Obs.	138011	138009	138008			
	T /2	E U /O /				
	Log (.	Energy Use/Out	tput)			
Post*Acquired	-0.392***	-0.443***	-0.590***			
\mathbf{D} (:(1:))	(0.129)	(0.149)	(0.178)			
R-sq. (within)	0.643	0.642	0.625			
NO. OI UDS.	138011	138009	138008			
	Log (Ce	$O_2 \text{ Emissions}/O$	utput)			
Post*Acquired	-0.318***	-0.380***	-0.519^{***}			
	(0.117)	(0.134)	(0.165)			
R-sq. (within)	0.653	0.653	0.629			
No. of Obs.	138010	138008	138007			

 Table E.1: Difference-in-differences estimates: Inverse probability of treatment weighting.

Note: The table shows the results of estimating equation 4 on the full dataset using inverse propensity of treatment weighting. The dependent variables are as listed in each panel. Each column in each panel is a separate regression for a particular outcome variable covering two time periods: the year relative to acquisition (as listed in each column) and a year before foreign-owned plants were acquired. Heteroskedasticity-robust standard errors clustered at the plant level are in parentheses. Year and plant fixed-effects are included in all specifications. *, ***, *** indicate statistical significance at 0.10, 0.05, and 0.01 level, respectively.

	Acquisition Year	1 Year Later	2 Years Later	3 Years Later	4 Years Later	5 Years Later	
	Log(Output)						
Post*Acquired	0.872^{***}	0.897^{***}	0.911***	1.079***	1.048^{***}	0.988^{***}	
	(0.104)	(0.111)	(0.110)	(0.119)	(0.121)	(0.130)	
R-sq. (within)	0.174	0.162	0.189	0.224	0.217	0.193	
No. of Obs.	1050	1050	1050	1050	1050	1050	
	Log (Energy Expenditure)						
Post*Acquired	0.510^{***}	0.550^{***}	0.577***	0.575***	0.465^{***}	0.623^{***}	
	(0.115)	(0.128)	(0.125)	(0.142)	(0.143)	(0.149)	
R-sq. (within)	0.089	0.080	0.105	0.091	0.091	0.115	
No. of Obs.	1024	1024	1024	1024	1024	1024	
	Log (Energy Expenditure/Output)						
Post*Acquired	-0.353***	-0.335***	-0.320***	-0.467***	-0.549^{***}	-0.341***	
	(0.101)	(0.104)	(0.109)	(0.113)	(0.117)	(0.117)	
R-sq. (within)	0.024	0.020	0.017	0.037	0.045	0.018	
No. of Obs.	1024	1024	1024	1024	1024	1024	
		Log	g (Energy Expen	diture/Materials)		
Post*Acquired	-0.388***	-0.382***	-0.435***	-0.506***	-0.606***	-0.416***	
	(0.113)	(0.116)	(0.130)	(0.135)	(0.143)	(0.145)	
R-sq. (within)	0.026	0.026	0.028	0.032	0.041	0.020	
No. of Obs.	962	962	962	962	962	962	

Table E.2: Regression results using extended data (1983-2008).

Note: The table reflects the result of estimating equation 4 on the matched sample described in Section 4 but extending the period up to 2008. The dependent variables are as listed in each panel. Expenditure on energy and materials and output are deflated using consumer price index (2014=100). Each column in each panel is a separate regression for a particular outcome variable covering two time periods: the year relative to acquisition (as listed in each column) and a year before foreign-owned plants were acquired. Heteroskedasticity-robust standard errors are in parentheses. *Post* dummy and plant fixed effects are included in all specifications. *, **, *** indicate statistical significance at 0.10, 0.05, and 0.01 level, respectively.

Appendix F Extended data: 1983-2008

	Matched sample				
Variables	(pre-acquisition, N=816)				
	Treated	Control	p-value		
Used in matching					
$Log (Output)_{t-1}$	12.58	12.55	0.869		
$Log (Energy expenditure/output)_{t-1}$	-3.70	-3.66	0.626		
$Log (Output)_{t-2}$	12.43	12.51	0.519		
$Log (Energy expenditure/output)_{t-2}$	-3.69	-3.70	0.915		
Not used in matching					
$Log (Energy expenditure)_{t-1}$	8.88	8.90	0.871		
$Log(Energy exp./materials exp.)_{t-1}$	-2.79	-2.90	0.317		
Δ Log (Energy expenditure) _{t-1}	0.14	0.09	0.461		
Δ Log (Energy expenditure/output) _{t-1}	-0.01	0.05	0.334		
Δ Log(Energy exp./materials exp.) _{t-1}	-0.04	0.01	0.407		

Table F.1: Balancing tests between domestic and acquired plants: Extended data (1983-2008).

Note: Acquired and treated plants are those that were previously domestic and remained acquired for at least two years after the acquisition case. Domestic plants are those plants that had no acquisition case, while control plants are those year-plant observations that had foreign equity below the threshold for four consecutive years.

Source: Indonesian Census of Manufacturing

	Acquisition Year	1 Year Later	2 Years Later		
	Log(Output)				
Post*Acquired	1.355^{***}	1.473^{***}	1.625^{***}		
	(0.133)	(0.150)	(0.141)		
R-sq. (within)	0.814	0.818	0.825		
No. of Obs.	229790	229790	229790		
	Log (Energy Expenditure)				
Post*Acquired	1.156^{***}	1.256^{***}	1.305^{***}		
	(0.138)	(0.157)	(0.156)		
R-sq. (within)	0.800	0.787	0.792		
No. of Obs.	228842	228842	228844		
	Log (Energ	gy Expenditure,	/Output)		
Post*Acquired	-0.170*	-0.218**	-0.315^{***}		
	(0.091)	(0.103)	(0.112)		
R-sq. (within)	0.667	0.655	0.648		
No. of Obs.	228840	228840	228842		

Table F.2: Regression results using inverse probability of treatmentweighting: Extended data (1983-2008).

Note: The table shows the results of estimating equation 4 on the extended dataset (1983-2008) using inverse propensity of treatment weighting. The dependent variables are as listed in each panel. Each column in each panel is a separate regression for a particular outcome variable covering two time periods: the year relative to acquisition (as listed in each column) and a year before foreign-owned plants were acquired. Heteroskedasticity-robust standard errors clustered at the plant level are in parentheses. Year and plant fixed-effects are included in all specifications. *, **, *** indicate statistical significance at 0.10, 0.05, and 0.01 level, respectively.

	Price-based measures			Physical Units			
	Acquisition Year	1 Year Later	2 Years Later	Acquisition Year	1 Year Later	2 Years Later	
	Log	Fotal Fuel in R	os)	Log(To	tal Fuel in MB	ГUs)	
Post*Acquired	0.392^{***}	0.596^{***}	0.543^{***}	0.343**	0.513^{***}	0.547^{***}	
	(0.149)	(0.158)	(0.172)	(0.165)	(0.173)	(0.189)	
R-sq. (within)	0.044	0.069	0.058	0.028	0.045	0.045	
No. of Obs.	812	815	805	812	815	806	
	Log(Ne	t Electricity in	Rps)	Log(Net 1	Electricity in M	BTUs)	
Post*Acquired	0.761^{***}	0.775^{***}	0.696^{***}	0.781^{***}	0.818^{***}	0.679^{***}	
	(0.204)	(0.206)	(0.217)	(0.201)	(0.208)	(0.219)	
R-sq. (within)	0.092	0.115	0.118	0.099	0.137	0.142	
No. of Obs.	714	713	711	714	713	711	
	Log(T	otal Fuel/Outp	ut)	Log(Total Fuel/Output)			
Post*Acquired	-0.422***	-0.452***	-0.429***	-0.471***	-0.535***	-0.428**	
	(0.148)	(0.153)	(0.160)	(0.164)	(0.171)	(0.182)	
R-sq. (within)	0.025	0.023	0.020	0.026	0.027	0.017	
No. of Obs.	812	815	805	812	815	806	
	Log(Net	Electricity/Ou	tput)	Log(Net Electricity/Output)			
Post*Acquired	-0.103	-0.343*	-0.389*	-0.083	-0.300	-0.406*	
	(0.203)	(0.202)	(0.213)	(0.201)	(0.204)	(0.219)	
R-sq. (within)	-0.001	0.011	0.015	-0.001	0.015	0.025	
No. of Obs.	714	713	711	714	713	711	

Table F.3: Regression results on fuel and electricity expenditure, use and intensities.

Note: The table reflects the result of estimating equation 4 on the matched sample described in Section 4. The dependent variables are as listed in each panel. Each column in each panel is a separate regression for a particular outcome variable covering two time periods: the year relative to acquisition (as listed in each column) and a year before foreign-owned plants were acquired. Heteroskedasticity-robust standard errors are in parentheses. *Post* dummy and plant fixed effects are included in all specifications. *, **, *** indicate statistical significance at 0.10, 0.05, and 0.01 level, respectively.

Appendix G Energy Prices, Scale Economies and Other Outcomes

	A	1 3 7 T (
	Acquisition Year	1 Year Later	2 Years Later			
	Fuel price					
Post*Acquired	-0.080	0.021	-0.059			
	(0.090)	(0.014)	(0.063)			
R-sq. (within)	0.005	0.013	0.005			
No. of Obs.	812	815	806			
]	Log(Fuel Price)				
Post*Acquired	0.049	0.083	0.014			
	(0.057)	(0.054)	(0.057)			
R-sq. (within)	0.004	0.008	0.001			
No. of Obs.	812	815	805			
	Acquisition Year	1 Year Later	2 Years Later			
	Ι	Electricity Price				
Post*Acquired	-0.001	-0.003	0.006			
	(0.003)	(0.003)	(0.007)			
R-sq. (within)	0.001	0.026	0.004			
No. of Obs.	714	713	711			
	Log(Electricity Price)					
Acquired	-0.020	-0.043	0.016			
	(0.040)	(0.044)	(0.051)			
R-sq. (within)	0.002	0.038	0.045			
No. of Obs	714	713	711			

Table G.1: Difference-in-differences analysis on matched sample: Energy input prices in levels and logarithms.

Note: The table reflects the result of estimating equation 4 on the matched sample using PSM. The dependent variables are as listed in each panel. Each column in each panel is a separate regression for a particular outcome variable covering two time periods: the year relative to acquisition (as listed in each column) and a year before foreign-owned plants were acquired. Heteroskedasticity-robust standard errors clustered at the plant level are in parentheses. Year and plant fixed-effects are included in all specifications. *, **, *** indicate statistical significance at 0.10, 0.05, and 0.01 level, respectively.

Figure	G.1:	Predicted	energy	expenditure	and	output.



Note: The figures represent the estimated coefficient from regressing energy expenditure on foreign affiliate dummy (i.e. 1 if foreign-owned; 0 otherwise), lagged output and their interaction. Dashed lines represent 95-percent confidence intervals. Each variable is expressed in log.

	All Sample		Matched Sample	
	(1)	(2)	(3)	(4)
Post*Acquired	0.836**	2.334^{***}	0.997**	1.931^{***}
ln(Output)	$egin{array}{c} (0.329) \ 0.571^{***} \ (0.005) \end{array}$	(0.250)	(0.404) 0.621^{***} (0.040)	(0.462)
$\ln(\text{Output})_{t-1}$	(0.000)	0.272^{***}	(01010)	0.250^{***}
Post*Acquired*ln(Output)	-0.060**	(0.000)	-0.086^{**}	(0.040)
$\operatorname{Post}^{*}\operatorname{Acquired}^{*}\ln(\operatorname{Output})_{t-1}$	(0.030)	-0.176^{***} (0.022)	(0.038)	-0.144^{***} (0.043)
Firm fixed effect	Yes	Yes	Yes	Yes
Year-fixed effect	Yes	Yes	Yes	Yes
R-sq. (within)	0.261	0.097	0.389	0.134
No. of Obs.	255450	228733	2994	2571

Table G.2: Dependent Variable: Log(Energy Expenditure)

Note: The table reflects the result of regressing the outcome variable on the unmatched sample. Each column in each panel is a separate regression. Heteroskedasticity-robust standard errors are in parentheses. *, **, *** indicate statistical significance at 0.10, 0.05, and 0.01 level, respectively.

	Acquisition Year	1 Year Later	2 Years Later			
	Log(Capital)					
Post*Acquired	0.622^{***}	0.738***	0.794^{***}			
	(0.147)	(0.178)	(0.208)			
R-sq. (within)	0.109	0.099	0.082			
No. of Obs.	658	644	627			
	Le	og(Employment))			
Post*Acquired	0.319^{***}	0.349^{***}	0.361^{***}			
	(0.050)	(0.055)	(0.061)			
R-sq. (within)	0.153	0.152	0.109			
No. of Obs.	840	840	840			
	Log(Capital-Labor Ratio)					
Post*Acquired	0.349**	0.349** 0.406**				
	(0.145)	(0.174)	(0.201)			
R-sq. (within)	0.034	0.030	0.030			
No. of Obs.	658	644	627			
	Log(Investment in Machinery)					
Post*Acquired	0.745^{***}	0.729^{***}	0.861^{***}			
	(0.178)	(0.202)	(0.245)			
R-sq. (within)	0.087	0.070	0.067			
No. of Obs.	650	637	620			
	Log(Total Investment, 5 categories)					
Post*Acquired	0.324	0.449	-0.062			
	(0.348)	(0.391)	(0.377)			
R-sq. (within)	0.006	0.008	0.000			
No. of Obs.	489	460	444			

Table G.3: PSM-DID estimates on capital, employment and investments.

Note: The table reflects the result of estimating equation 4 on the matched sample using PSM. The dependent variables are as listed in each panel. Each column in each panel is a separate regression for a particular outcome variable covering two time periods: the year relative to acquisition (as listed in each column) and a year before foreign-owned plants were acquired. Heteroskedasticity-robust standard errors clustered at the plant level are in parentheses. Year and plant fixed-effects are included in all specifications. *, **, *** indicate statistical significance at 0.10, 0.05, and 0.01 level, respectively.

Appendix H Analysis of Channels: Additional Table

	Un	matched san	nple	Matched Sample (HHI<5%)		
Variables	(162 acquired vs 1,271 domestic)			(58 treated and 58 controls)		
	Acquired	Domestic	p-value	Treated	Control	p-value
Used in matching						
$\text{Log (Real output)}_{t-1}$	12.65	0.49	0.000	12.38	12.31	0.855
$Log (Energy expenditure/output)_{t-1}$	-3.97	0.00	0.000	-3.56	-3.49	0.777
$\text{Log (Real output)}_{t-2}$	12.79	0.00	0.000	12.10	12.27	0.649
Log (Energy expenditure/output) _{$t-2$}	-3.92	0.00	0.003	-3.51	-3.47	0.851
Unused in matching						
$Log (Energy expenditure)_{t-1}$	8.69	0.00	0.000	8.82	8.82	0.986
Log (predicted energy intensity, $\operatorname{continuous})_{t-1}$	-3.43	0.00	0.000	-3.45	-3.15	0.042
Log (predicted energy intensity, percentile) $_{t-1}$	3.82	0.00	0.001	3.85	4.01	0.089
$Log (Employment)_{t-1}$	5.24	0.00	0.000	4.82	4.90	0.743
Exporter $\operatorname{dummy}_{t-1}$	1.49	0.00	0.000	1.63	1.82	0.191
Share of imported materials t_{t-1}	0.27	0.00	0.000	0.27	0.14	0.028
Share of skilled workers $_{t-1}$	0.20	0.00	0.000	0.21	0.21	0.957
$Log(Investment in machinery)_{t-1}$	10.34	0.00	0.000	10.85	10.18	0.290
$Log(Energy exp./materials exp.)_{t-1}$	-3.16	0.00	0.000	-2.73	-2.67	0.831
Δ Log (Real output) _{t-1}	0.10	0.00	0.158	0.28	0.04	0.236
Δ Log (Energy expenditure) _{t-1}	0.11	0.00	0.872	0.22	0.02	0.351
Δ Log (Energy expenditure/output) _{t-1}	0.01	0.00	0.295	-0.05	-0.03	0.822
Δ Log(Energy exp./materials exp.) _{t-1}	-0.05	0.00	0.067	-0.22	0.01	0.174

Table H.1: Balancing test for matched domestic and acquired plants, ${\rm HHI}{<}5\%.$

Note: Acquired and treated plants are those that were previously domestic and remained acquired for at least two years after the acquisition case. Domestic plants are those plants that had no acquisition case, while control plants are those year-plant observations that had foreign equity below the threshold for four consecutive years. Source: Indonesian Census of Manufacturing

Appendix I Divestment: Additional Table

	Unmatched sample			Matched sample			
Variables	(597 Divested vs 42,084 Foreign)			(256 treated vs 256 controls)			
	Divested	Foreign	p-value	Treated	Control	p-value	
Used in matching							
$Log (Real output)_{t-1}$	9.56	7.91	0.00	10.94	10.96	0.87	
$Log (Energy expenditure/output)_{t-1}$	-4.04	-3.84	0.00	-4.21	-4.19	0.86	
$Log (Real output)_{t-2}$	9.57	7.96	0.00	10.83	10.76	0.60	
$Log (Energy expenditure/output)_{t-2}$	-3.97	-3.83	0.04	-4.13	-4.08	0.68	
Unused in matching							
$Log (Energy expenditure)_{t-1}$	5.60	4.15	0.00	6.72	6.76	0.79	
$Log(Energy use)_{t-1}$	8.51	7.11	0.00	9.66	9.69	0.80	
$Log (CO2 emission)_{t-1}$	12.91	11.49	0.00	14.05	14.08	0.82	
$Log (Employment)_{t-1}$	5.03	4.19	0.00	5.76	5.69	0.46	
Exporter $\operatorname{dummy}_{t-1}$	0.24	0.10	0.00	0.33	0.35	0.64	
Share of imported materials $_{t-1}$	0.26	0.10	0.00	0.30	0.36	0.05	
Share of skilled workers $_{t-1}$	0.20	0.14	0.00	0.22	0.21	0.39	
$Log(Investment in machinery)_{t-1}$	7.79	5.66	0.00	8.84	8.96	0.55	
$Log(Energy use/output)_{t-1}$	-1.13	-0.88	0.00	-1.28	-1.26	0.87	
$Log(CO2 \text{ emission/output})_{t-1}$	3.27	3.51	0.00	3.11	3.13	0.89	
$Log(Energy exp./materials exp.)_{t-1}$	-3.22	-3.00	0.00	-3.54	-3.37	0.19	
Δ Log (Real output) _{t-1}	0.10	0.05	0.13	0.11	0.19	0.24	
Δ Log (Energy expenditure) _{t-1}	0.03	0.06	0.49	0.03	0.08	0.51	
Δ Log (Energy use) _{t-1}	0.03	0.08	0.23	0.02	0.10	0.40	
Δ Log (CO2 emission) _{t-1}	0.03	0.08	0.23	0.02	0.10	0.43	
Δ Log (Energy expenditure/output) _{t-1}	-0.06	0.01	0.09	-0.09	-0.11	0.78	
Δ Log(Energy use/output) _{t-1}	-0.07	0.02	0.03	-0.09	-0.09	0.95	
$\Delta \text{ Log}(\text{CO2 emission/output})_{t-1}$	-0.07	0.03	0.03	-0.09	-0.10	0.91	
Δ Log(Energy exp./materials exp.) _{t-1}	-0.05	0.01	0.21	-0.11	-0.13	0.77	

Table I.1: Balancing tests for matched and unmatched divested plants.

Source of raw data: Indonesian Census of Manufacturing.